Lawrence Berkeley National Laboratory

Lawrence Berkeley National Laboratory

Title

Moments for general quadratic densities in n dimensions

Permalink

https://escholarship.org/uc/item/2c4535mg

Author

Furman, Miguel A.

Publication Date 2002-03-20

Moments for General Quadratic Densities in n Dimensions^{*}

Miguel A. Furman[†]

Center for Beam Physics Accelerator and Fusion Research Division Lawrence Berkeley National Laboratory, MS 71-259 Berkeley, CA 94720

December 18, 2002

Abstract

We present the calculation of the generating functions and the *r*th-order moments for densities of the form $\rho(\mathbf{x}) \propto g(s(\mathbf{x}))$ where g(s) is a non-negative function of the quadratic "action" $s(\mathbf{x}) = \sum_{i,j} H_{ij} x_i x_j$, where $\mathbf{x} = (x_1, x_2, \dots, x_n)$ is a real *n*-dimensional vector and *H* is a real, symmetric $n \times n$ matrix whose eigenvalues are strictly positive. In particular, we find the connection between the (r + 2)th-order and *r*th-order moments, which constitutes a generalization of the Gaussian moment theorem [1], which corresponds to the particular choice $g(s) = e^{-s/2}$. We present several examples for specific choices for g(s), including the explicit expression for the generating function for each case and the subspace projection of $\rho(\mathbf{x})$ in a few cases. We also provide the straightforward generalizations to: 1) the case where $g = g(s(\mathbf{x}) + \mathbf{a} \cdot \mathbf{x})$, where $\mathbf{a} = (a_1, a_2, \dots, a_n)$ is an arbitrary real *n*-dimensional vector, and 2) the complex case, in which the action is of the form $s(\mathbf{z}) = \sum_{i,j} H_{ij} z_i^* z_j$ where $\mathbf{z} = (z_1, z_2, \dots, z_n)$ is an *n*-dimensional complex vector and *H* is a Hermitian $n \times n$ matrix whose eigenvalues are strictly positive.

DISCLAIMER

This document was prepared as an account of work sponsored by the United States Government. While this document is believed to contain correct information, neither the United States Government nor any agency thereof, nor The Regents of the University of California, nor any of their employees, makes any warranty, express or implied, or assumes any legal responsibility for the accuracy, completeness, or usefulness of any information, apparatus, product, or process disclosed, or represents that its use would not infringe privately owned rights. Reference herein to any specific commercial product, process, or service by its trade name, trademark, manufacturer, or otherwise, does not necessarily constitute or imply its endorsement, recommendation, or favoring by the United States Government or any agency thereof, or The Regents of the University of California. The views and opinions of authors expressed herein do not necessarily state or reflect those of the United States Government or any agency thereof, or The Regents of the University of California.

Ernest Orlando Lawrence Berkeley National Laboratory is an equal opportunity employer.

^{*}Work supported by the US DOE under contract DE-AC03-76SF00098.

[†]mafurman@lbl.gov

Contents

1	Introduction.	3
2	Averages, Moments and Generating Functions.	4
3	Strictly Quadratic Density and Moments. 3.1 Recursion Formulas for the Moments. 3.1.1 Special Case: 2nd-order Moments.	5 7 8
4	General Quadratic Density and Moments.	8
5	Subspace Projection.	10
6	Examples. 6.1 Gaussian Density. 6.1.1 Subspace Projection for the Gaussian Density. 6.2 Gamma-Function Density. 6.3 Exponential Density. 6.4 Generalized Lorentzian Density. 6.4.1 Subspace Projection for the Lorentzian Density. 6.5 Microcanonical Density. 6.5.1 Subspace Projection for the Microcanonical Density. 6.6 Beta-Function Density. 6.6.1 Waterbag Density. 6.6.2 Generalized Waterbag Density.	13 13 14 14 14 15 15 16 16 16 16 16 17
7	Extension to the Complex Space.7.1 Recursion Formulas for the Moments.7.2 Example: the Complex Gaussian Density.7.3 Subspace Projection.	17 21 22 22
8	Conclusions.	23
\mathbf{A}	Certain <i>n</i> -Dimensional Integrals.	25
в	Complex Integrals.	26
\mathbf{L}	ist of Tables	

1	Summary of examples	3

1 Introduction.

The Gaussian density plays an ubiquitous role in many areas of physics. For example, consider a system of *n* degrees of freedom characterized by the real *n*-dimensional vector $\mathbf{x} = (x_1, x_2, \dots, x_n)$ and a quadratic "action" of the form $s(\mathbf{x}) = \sum_{i,j} H_{ij} x_i x_j$ where *H* is a real, symmetric, positive-definite matrix¹. A Gaussian density, given by $\rho(\mathbf{x}) \propto \exp(-s(\mathbf{x})/2)$, yields a 2nd-order moment $\langle x_i x_j \rangle = H_{ij}^{-1}$. Since the entire physical contents of the model is embodied in the matrix *H*, all higher-order moments are recursively expressed in terms of the 2nd-order moments by means of the well-known recursion formula

$$\langle \underbrace{x_i x_j x_k \cdots x_m}_{r \; x's} \rangle = \langle x_i x_j \rangle \langle \underbrace{x_k \cdots x_m}_{r-2 \; x's} \rangle + \text{permutations}$$
(1.1)

where the "permutations" are such as to make the right-hand side a fully symmetric function of the r indices i, j, k, \dots, m . Eq. (1.1) is isomorphic with Wick's expansion of the vacuum expectation value of a product of bosonic operators [2], and with Mayer's cluster expansion for a gas with two-body potentials [3]. By successive application of this recursion, one arrives at the Gaussian moment theorem [1], which expresses any even-order Gaussian moment purely in terms of 2nd-order moments. For example, the 4th-order moment is given by the fully symmetrized product of 2nd-order moments

$$\langle x_i x_j x_k x_m \rangle = \langle x_i x_j \rangle \langle x_k x_m \rangle + \langle x_i x_k \rangle \langle x_j x_m \rangle + \langle x_i x_m \rangle \langle x_j x_k \rangle \tag{1.2}$$

In beam physics it is sometimes the case that the particle distribution in phase space is not Gaussian but it is, nevertheless, matched to the lattice optics in linear order, and one may want to study the time evolution of the moments of the canonical coordinates (q_i, p_i) [4, 5]. The linearly-matched condition implies that the density is a function of the quadratic part of the Hamiltonian, hence it is of the form described above. Motivated by such cases, we consider in this article a more general density $\rho(\mathbf{x}) \propto g(s)$, where g(s) is any real, non-negative function of the quadratic action $s(\mathbf{x})$. As in the Gaussian case, the entire information of the model is contained in H, hence all moments must be expressible in terms of the 2nd-order moments $\langle x_i x_j \rangle$. Furthermore, since the *r*th-order moment is a fully symmetric tensor of r indices, and the only ingredients to construct it are the $\langle x_i x_j \rangle$'s, it must be the case that a higher-order moment must proportional to the fully symmetrized product of these. Indeed, in this article we show that, in this more general case, the main expressions are given by

$$\langle x_i x_j \rangle = a H_{ij}^{-1} \tag{1.3a}$$

$$\langle x_i x_j x_k \cdots x_m \rangle = b \left[\langle x_i x_j \rangle \langle x_k \cdots x_m \rangle + \text{permutations} \right]$$
 (1.3b)

$$= c \left[\langle x_i x_j \rangle \langle x_k x_m \rangle \langle \cdots \rangle \cdots + \text{permutations} \right]$$
(1.3c)

The main purpose of this article is the calculation of the coefficients a, b, and c given the dimensionality n, the order of the moment r, and the function g(s) (the Gaussian density is unique in that a = b = c = 1 for all n and r). Although in beam physics [4, 5] the only interesting cases are $1 \le n \le 6$, the calculation is just as simple for any value of n, hence we allow it to be an arbitrary number whose only requirement is n > 0. We also compute the generating functions for the moments, and present a discussion of the subspace projection of $\rho(\mathbf{x})$, which is useful when averaging a function that depends only on a subset of the components of \mathbf{x} .

In Sec. 2 we set up the general formalism. In Sec. 3 we consider the strictly quadratic action $s(\mathbf{x}) = \sum_{i,j} H_{ij}x_ix_j$. In Sec. 4 we extend the analysis to the general quadratic action of the form $\sum_{i,j} H_{ij}x_ix_j + \sum_i a_ix_i$, where (a_1, a_2, \dots, a_n) is an arbitrary real *n*-dimensional vector. Section 5 describes the subspace projection of a distribution. Section 6 contains results for specific choices of the function g(s). Section 7 contains the generalization of the above results to densities $\rho(\mathbf{z})$ defined over the complex plane whose points are described by the complex *n*-dimensional vector \mathbf{z} . Section 8 summarizes our conclusions. Appendix A contains the definition and basic properties of spherical coordinates in *n* dimensions, along with some other mathematical details, and Appendix B extends some of these results to the complex case.

¹A positive-definite matrix is, by definition, one whose all eigenvalues are real and strictly > 0.

We use throughout Dirac's "bra-ket" notation [6] interchangeably with the conventional notation for the matrix and scalar product. Thus, for example, if **a** and **b** represent two real *n*-dimensional vectors with components (a_1, \dots, a_n) and (b_1, \dots, b_n) , respectively, then the scalar product $\langle \mathbf{a}|M|\mathbf{b}\rangle$ stands for $\sum_{i,j} M_{ij}a_ib_j$, where M is an arbitrary real $n \times n$ matrix. If **u** and **v** represent complex vectors, then $\langle \mathbf{u}|M|\mathbf{v}\rangle$ stands for $\sum_{i,j} M_{ij}u_i^*v_j$ where M is an arbitrary complex $n \times n$ matrix (we use interchangeably the asterisk and the overbar to denote complex conjugation).

2 Averages, Moments and Generating Functions.

Let $\rho(\mathbf{x})$ be a real function of the *n*-dimensional vector \mathbf{x} such that $\rho(\mathbf{x}) \ge 0$ for all \mathbf{x} . We assume that $\rho(\mathbf{x})$ is normalizable and we chose, without any loss of generality, unit normalization, *i.e.*

$$\int d^n \mathbf{x} \,\rho(\mathbf{x}) = 1 \tag{2.1}$$

The average of any function $f(\mathbf{x})$ with respect to the density $\rho(\mathbf{x})$ is given by

$$\langle f(\mathbf{x}) \rangle = \int d^n \mathbf{x} \, \rho(\mathbf{x}) f(\mathbf{x})$$
 (2.2)

The rth-order moment is, by definition, the average of the product of r components of \mathbf{x} ,

$$C_{ij\cdots}^{(r)} \equiv \langle x_i x_j \cdots \rangle = \int d^n \mathbf{x} \,\rho(\mathbf{x})(x_i x_j \cdots) \qquad (r \text{ indices})$$
(2.3)

with $r = 1, 2, \cdots$. If $\rho(\mathbf{x})$ is nonzero only over a finite region of \mathbf{x} , then obviously the moments exist for all r. If, on the other hand, $\rho(\mathbf{x})$ extends over all space and vanishes as $x \to \infty$ as $\rho(\mathbf{x}) \sim x^{-2p}$, where $x = |\mathbf{x}|$ and p > 0, then the *r*th-order moment exists only if r < 2p - n. In particular, the normalizability property of $\rho(\mathbf{x})$ requires 2p > n. If $\rho(\mathbf{x})$ vanishes at infinity faster than any power of x, the moments exist for all r. If $\rho(\mathbf{x})$ is of even parity, *i.e.*, $\rho(-\mathbf{x}) = \rho(\mathbf{x})$, then all odd-order (r = odd) moments vanish.

The $C^{(r)}$'s are fully symmetric tensors whose generating function is the Fourier transform of $\rho(\mathbf{x})$,

$$\mathcal{G}(\mathbf{k}) = \int d^n \mathbf{x} \, \rho(\mathbf{x}) \, e^{-i\mathbf{k} \cdot \mathbf{x}} = \left\langle e^{-i\mathbf{k} \cdot \mathbf{x}} \right\rangle \tag{2.4}$$

hence

$$C_{ij\cdots}^{(r)} = i^r \left. \frac{\partial^r \mathcal{G}(\mathbf{k})}{\partial k_i \partial k_j \cdots} \right|_{\mathbf{k}=0} \qquad (r \text{ indices})$$
(2.5)

Note that $\mathcal{G}(\mathbf{k})$ exists for all \mathbf{k} provided only that $\rho(\mathbf{x})$ vanishes as $x \to \infty$ faster than x^{-n} . However, as seen above, the existence of the moments requires a faster falloff of $\rho(\mathbf{x})$ at ∞ . In other words, although $\mathcal{G}(\mathbf{k})$ itself exists, its *r*th derivative at $\mathbf{k} = 0$ is finite only if r < 2p - n.

The normalization condition (2.1) implies $\mathcal{G}(0) = 1$. In case that all moments exist, we can expand $e^{-i\mathbf{k}\cdot\mathbf{x}}$ in (2.4) and integrate term by term to obtain

$$\mathcal{G}(\mathbf{k}) = \int d^{n} \mathbf{x} \rho(\mathbf{x}) \left(1 - i \sum_{i} k_{i} x_{i} + \frac{(-i)^{2}}{2!} \sum_{i,j} k_{i} k_{j} x_{i} x_{j} + \cdots \right)$$

= $1 - i \sum_{i} k_{i} C_{i}^{(1)} + \frac{(-i)^{2}}{2!} \sum_{i,j} k_{i} k_{j} C_{ij}^{(2)} + \cdots$ (2.6)

If we express $f(\mathbf{x})$ in terms of its Fourier transform,

$$f(\mathbf{x}) = \int \frac{d^n \mathbf{k}}{(2\pi)^n} e^{i\mathbf{k}\cdot\mathbf{x}} \tilde{f}(\mathbf{k})$$
(2.7)

then we also have

$$\langle f(\mathbf{x}) \rangle = \int \frac{d^n \mathbf{k}}{(2\pi)^n} \left\langle e^{i\mathbf{k}\cdot\mathbf{x}} \right\rangle \tilde{f}(\mathbf{k}) = \int \frac{d^n \mathbf{k}}{(2\pi)^n} \mathcal{G}(-\mathbf{k}) \tilde{f}(\mathbf{k})$$
(2.8)

3 Strictly Quadratic Density and Moments.

Given a real symmetric $n \times n$ matrix H, we consider the quadratic form $s(\mathbf{x})$

$$s(\mathbf{x}) = \langle \mathbf{x} | H | \mathbf{x} \rangle \equiv \sum_{i,j=1}^{n} H_{ij} x_i x_j$$
(3.1)

where \mathbf{x} is an arbitrary real *n*-dimensional vector. Then we can state the following theorem [7,8]:

Theorem 3.1 $s(\mathbf{x}) > 0$ for all $\mathbf{x} \neq 0$ if and only if all eigenvalues of H are strictly > 0 (i.e., H is a positive-definite matrix). Furthermore, $s(\mathbf{x}) = 0$ if and only if $\mathbf{x} = 0$.

Proof. Since *H* is real and symmetric, it has *n* real eigenvalues and *n* nontrivial orthogonal eigenvectors, called h_k and \mathbf{v}_k , respectively, with $k = 1, 2, \dots, n$. Choosing $\mathbf{x} = \mathbf{v}_k$ yields $s(\mathbf{x}) = h_k \mathbf{v}_k^2$. Since $s(\mathbf{x}) > 0$ and \mathbf{v}_k is nontrivial, one obtains $h_k > 0$ for any *k*. For the proof of the converse theorem, we assume $h_k > 0$ for all *k* and use the fact that the \mathbf{v}_k 's form a complete orthogonal set to express \mathbf{x} as a linear combination $\mathbf{x} = \sum_i a_i \mathbf{v}_i$, hence

$$s(\mathbf{x}) = \sum_{i,j=1}^{n} a_i a_j \langle \mathbf{v}_i | H | \mathbf{v}_j \rangle = \sum_{i=1}^{n} a_i^2 h_i \mathbf{v}_i^2$$
(3.2)

which is > 0 provided at least one of the a_i 's is $\neq 0$. Setting $s(\mathbf{x}) = 0$ in this equation yields $a_i = 0$ for all i, which corresponds to $\mathbf{x} = 0$.

Corollary 1: Define a "submatrix" H' of H to be the matrix obtained from H by eliminating an arbitrary subset of columns and the corresponding subset of rows. By construction, H' is a symmetric matrix hence the above theorem applies to H' as well. This corollary is proven by choosing vectors \mathbf{x} whose components corresponding to the eliminated rows and columns are 0. This implies that any submatrix of a positive-definite matrix. In particular, by eliminating all but one of the columns and the corresponding rows, this corollary implies that the diagonal elements of H are > 0.

Corollary 2: The above theorem and corollary apply to H^{-1} as well. This follows from the obvious fact that H^{-1} exists (because det H > 0) and is symmetric. In particular, this implies that the inverse of a positive-definite matrix is itself a positive-definite matrix.

We now consider a density function $\rho(\mathbf{x})$ that depends on \mathbf{x} only through $s(\mathbf{x})$, *i.e.*,

$$\rho(\mathbf{x}) = Ng(s(\mathbf{x})) \tag{3.3}$$

where g(s) is a real function of s such that $g(s) \ge 0$ for $0 \le s \le \infty$, and N is a normalization constant determined by (2.1),

$$N^{-1} = \int d^{n} \mathbf{x} \, g(\langle \mathbf{x} | H | \mathbf{x} \rangle) \tag{3.4}$$

Now *H* can be diagonalized by a matrix U, $H = U^{-1}H_DU$ where $H_D = \text{diag}(h_1, h_2, \dots, h_n)$. Since *H* is symmetric, *U* is orthogonal, *i.e.* $U^{-1} = U^T$. Since $h_i > 0$ for all *i*, the matrix $H_D^{1/2}$ exists unambiguously hence we can define the change of variables $\mathbf{v} = H_D^{1/2}U\mathbf{x}$ yielding $s(\mathbf{x}) = \langle \mathbf{v} | \mathbf{v} \rangle = v^2$. Using the integrals in Appendix A one obtains

$$N^{-1} = \frac{1}{\sqrt{\det H}} \int d^n \mathbf{v} \, g(v^2) = \frac{\Omega_n g_0}{\sqrt{\det H}}$$
(3.5)

where $\Omega_n = 2\pi^{n/2}/\Gamma(n/2)$ is the area of the *n*-dimensional unit sphere and

$$g_0 \equiv \int_0^\infty dv \, g(v^2) v^{n-1} = \frac{1}{2} \int_0^\infty ds \, g(s) s^{n/2 - 1}$$
(3.6)

Using the positivity of $s(\mathbf{x})$ (Theorem 3.1) one can write (3.3) in the form

$$\rho(\mathbf{x}) = N \int_{0}^{\infty} ds \, g(s) \, \delta(\langle \mathbf{x} | H | \mathbf{x} \rangle - s)$$
(3.7)

Using again the results in Appendix A and the change of variable $s = v^2$ one obtains

$$\mathcal{G}(\mathbf{k}) = N \int_{0}^{\infty} ds \, g(s) \int d^{n} \mathbf{x} \, \delta(\langle \mathbf{x} | H | \mathbf{x} \rangle - s) e^{-i\mathbf{k} \cdot \mathbf{x}}$$
$$= \frac{\Gamma(n/2)}{g_{0}} \left(\frac{2}{q}\right)^{\nu} \int_{0}^{\infty} dv \, g(v^{2}) v^{n/2} J_{\nu}(qv), \qquad \nu = n/2 - 1$$
(3.8)

where $J_{\nu}(z)$ is the ordinary Bessel function of order ν , and **q** is a vector defined by $\mathbf{q} = H_D^{-1/2} U \mathbf{k}$ whose magnitude is given by

$$q^{2} = \langle \mathbf{q} | \mathbf{q} \rangle = \langle \mathbf{k} | H^{-1} | \mathbf{k} \rangle \tag{3.9}$$

If $\rho(\mathbf{x})$ falls off as $x \to \infty$ faster than any power of x, then $g(v^2)$ falls off faster than any power of v, hence we can expand the Bessel function in Eq. (3.8) in its Taylor series and integrate term by term to yield

$$\mathcal{G}(\mathbf{k}) = \sum_{\ell=0}^{\infty} A(n;\ell) \frac{(-q^2/2)^{\ell}}{\ell!}$$
(3.10)

where $A(n; \ell)$ is given by

$$A(n;\ell) = \frac{g_{\ell}/g_0}{2^{\ell}(n/2)_{\ell}}$$
(3.11)

where the symbol $(z)_{\ell}$ is defined by [9]

$$(z)_{\ell} = \frac{\Gamma(z+\ell)}{\Gamma(z)} = \begin{cases} 1, & \ell = 0\\ z(z+1)\cdots(z+\ell-1), & \ell \ge 1 \end{cases}$$
(3.12)

and where g_{ℓ} is a moment of the function g(s),

$$g_{\ell} \equiv \int_{0}^{\infty} dv \, g(v^2) v^{n+2\ell-1} = \frac{1}{2} \int_{0}^{\infty} ds \, g(s) s^{n/2+\ell-1}$$
(3.13)

Defining $G \equiv H^{-1}$ and writing q^2 explicitly in the right-hand side of (3.10) and comparing with (2.6) taking into account the symmetry² of G yields

$$C_{ij\cdots}^{(r)} = \begin{cases} A(n;\ell)S_{ij\cdots}^{(r)} & \text{for } r = 2\ell \text{ indices} \\ 0 & \text{for } r = \text{odd} \end{cases}$$
(3.14)

 $^{{}^{2}}G$ is symmetric because H is symmetric.

where $S_{ij...}^{(r)}$ is, by definition, the fully symmetric tensor of rank r formed out of products of G's. This tensor can be defined recursively by

$$S_{ijkm\cdots}^{(r)} = G_{ij}S_{km\cdots}^{(r-2)} + \text{permutations}$$
(3.15)

with $S^{(0)} = 1$. For example,

$$S_{ij}^{(2)} = G_{ij} (3.16a)$$

$$S_{ijkm}^{(4)} = G_{ij}G_{km} + G_{ik}G_{jm} + G_{im}G_{jk}$$
(3.16b)

etc. The tensor $S^{(r)}$, with $r = 2\ell$ indices, has $(2\ell)!/2^{\ell}\ell! = 1 \cdot 3 \cdot 5 \cdots (2\ell - 1)$ monomials, each formed out of a product of ℓ G's. Alternatively, it is straightforward to prove that

$$\frac{\frac{\partial^r \langle \mathbf{k} | G | \mathbf{k} \rangle^{\ell}}{\partial k_i \partial k_j \cdots}}{\sum_{r \ k's}} \bigg|_{\mathbf{k}=0} = \begin{cases} 2^{\ell} \ell! \, S_{ij\cdots}^{(r)} & \text{for } r = 2\ell \\ 0 & \text{otherwise} \end{cases}$$
(3.17)

which, when combined with definition (2.5) applied to the formula (3.10), yields (3.14).

If $\rho(\mathbf{x})$ vanishes as $x \to \infty$ as x^{-2p} then $C^{(r)}$ exists and is given by expression (3.14) only as long as g_{ℓ} exists, *i.e.*, only for r < 2p - n.

3.1 Recursion Formulas for the Moments.

Note that Eqs. (3.14) and (3.16) imply that all moments with $\ell > 1$ are determined by the second-order $(\ell = 1)$ moments as long as g_{ℓ} exists. Thus Eq. (3.14) implies the fundamental recursion relation

$$C_{ijkm\cdots}^{(r)} = B(n;\ell) \left[C_{ij}^{(2)} C_{km\cdots}^{(r-2)} + \text{permutations} \right]$$
(3.18)

with $C^{(0)} = 1$ or, equivalently,

$$\langle x_i x_j x_k x_m \cdots \rangle = B(n; \ell) \left[\langle x_i x_j \rangle \langle x_k x_m \cdots \rangle + \text{permutations} \right]$$
(3.19)

where

$$B(n;\ell) = \frac{A(n;\ell)}{A(n;\ell-1)A(n;1)} = \frac{g_{\ell}g_0}{g_1g_{\ell-1}} \cdot \frac{n}{n+2(\ell-1)}$$
(3.20)

By recursive applications of Eq. (3.18) to its right-hand side, one can reach an expression for $C^{(r)}$ purely in terms of $C^{(2)}$'s,

$$C_{ijkm\cdots}^{(r)} = D(n;\ell) \Big[\underbrace{C_{ij}^{(2)}C_{km}^{(2)}\cdots}_{\ell \ C^{(2)}'s} + \text{permutations}\Big]$$
(3.21)

or, equivalently,

$$\langle x_i x_j x_k x_m \cdots \rangle = D(n; \ell) \left[\underbrace{\langle x_i x_j \rangle \langle x_k x_m \rangle \cdots}_{\ell \text{ factors}} + \text{permutations} \right]$$
(3.22)

where

$$D(n;\ell) = \frac{A(n;\ell)}{(A(n;1))^{\ell}} = \frac{g_{\ell}g_0^{\ell-1}(n/2)^{\ell}}{g_1^{\ell}(n/2)_{\ell}}$$
(3.23)

It is straightforward to verify that A(n; 0) = B(n; 1) = D(n; 1) = 1, as consistency demands. As an example we consider the 4th-order moments by setting $\ell = 2$. In this case we have

$$D(n;2) = B(n;2) = \frac{n g_2 g_0}{(n+2) g_1^2}$$
(3.24)

hence

$$C_{ijkm}^{(4)} = \frac{n g_2 g_0}{(n+2) g_1^2} \left[C_{ij}^{(2)} C_{km}^{(2)} + C_{ik}^{(2)} C_{jm}^{(2)} + C_{im}^{(2)} C_{jk}^{(2)} \right]$$
(3.25)

where

$$C_{ij}^{(2)} = \frac{g_1}{ng_0} G_{ij} \tag{3.26}$$

Equations (3.10–3.16), complemented by Eqs. (3.18–3.23), constitute the central results of this article.

3.1.1 Special Case: 2nd-order Moments.

Eq. (3.26) and the definition of G can be cast in matrix form,

$$G = H^{-1} = \frac{ng_0}{g_1} C^{(2)} \tag{3.27}$$

where $C^{(2)}$ is the "correlation matrix"

$$C^{(2)} = \begin{pmatrix} \langle x_1^2 \rangle & \langle x_1 x_2 \rangle & \cdots & \langle x_1 x_n \rangle \\ \langle x_1 x_2 \rangle & \langle x_2^2 \rangle & \cdots & \langle x_2 x_n \rangle \\ \vdots & \vdots & \ddots & \vdots \\ \langle x_1 x_n \rangle & \langle x_2 x_n \rangle & \cdots & \langle x_n^2 \rangle \end{pmatrix}$$
(3.28)

which is, obviously, symmetric. Since H is positive-definite, Corollary 2 implies that so is $C^{(2)}$.

For the specific case n = 2 we obtain

$$\det G = \left(\frac{2g_0\epsilon}{g_1}\right)^2 \tag{3.29}$$

where we have defined

$$\epsilon^2 = \langle x_1^2 \rangle \langle x_2^2 \rangle - \langle x_1 x_2 \rangle^2 \tag{3.30}$$

which is guaranteed to be positive by Schwarz's inequality. The matrix H is, therefore,

$$H = G^{-1} = \frac{g_1}{2g_0\epsilon^2} \begin{pmatrix} \langle x_2^2 \rangle & -\langle x_1 x_2 \rangle \\ -\langle x_1 x_2 \rangle & \langle x_1^2 \rangle \end{pmatrix}$$
(3.31)

with eigenvalues

$$h_{\pm} = \frac{g_1}{4g_0\epsilon^2} \left[\langle x_1^2 \rangle + \langle x_2^2 \rangle \pm \sqrt{(\langle x_1^2 \rangle - \langle x_2^2 \rangle)^2 + 4\langle x_1 x_2 \rangle^2} \right]$$
(3.32)

Note that $h_{\pm} > 0$, as it should be the case.

4 General Quadratic Density and Moments.

We now consider a density function $\rho(\mathbf{x})$ that depends on \mathbf{x} through the combination

$$\rho(\mathbf{x}) = Ng(\langle \mathbf{x} | H | \mathbf{x} \rangle + \langle \mathbf{x} | \mathbf{a} \rangle) \tag{4.1}$$

where H is a positive-definite symmetric matrix, **a** is a constant vector and $g \ge 0$ for all **x**. The normalization constant N is determined by

$$N^{-1} = \int d^{n} \mathbf{x} \, g(\langle \mathbf{x} | H | \mathbf{x} \rangle + \langle \mathbf{x} | \mathbf{a} \rangle)$$
(4.2)

The linear term can be eliminated by a shift of origin in \mathbf{x} defined by $\mathbf{x} = \mathbf{y} + \mathbf{h}$ where \mathbf{h} is a constant vector to be determined. Using the symmetry property of H we obtain

$$\langle \mathbf{x}|H|\mathbf{x}\rangle + \langle \mathbf{x}|\mathbf{a}\rangle = \langle \mathbf{y}|H|\mathbf{y}\rangle + 2\langle \mathbf{y}|H|\mathbf{h}\rangle + \langle \mathbf{h}|H|\mathbf{h}\rangle + \langle \mathbf{y}|\mathbf{a}\rangle + \langle \mathbf{h}|\mathbf{a}\rangle$$
(4.3)

It is clear that the term linear in **y** is eliminated by the choice $2H\mathbf{h} + \mathbf{a} = 0$, thus $\mathbf{h} = -H^{-1}\mathbf{a}/2$ hence

$$\langle \mathbf{x}|H|\mathbf{x}\rangle + \langle \mathbf{x}|\mathbf{a}\rangle = \langle \mathbf{y}|H|\mathbf{y}\rangle + \Delta$$
 (4.4)

where $\Delta = -\frac{1}{4} \langle \mathbf{a} | G | \mathbf{a} \rangle$ and $G \equiv H^{-1}$, hence

$$N^{-1} = \int d^{n} \mathbf{y} \, g(\langle \mathbf{y} | H | \mathbf{y} \rangle + \Delta) \tag{4.5}$$

Following the same steps as in Sec. 3, using the results from Appendix A and defining the change of variables $\mathbf{y} = U^{-1} H_D^{-1/2} \mathbf{v}$, one obtains

$$N^{-1} = \frac{1}{\sqrt{\det H}} \int d^n \mathbf{v} \, g(v^2 + \Delta) = \frac{\Omega_n g_0(\Delta)}{\sqrt{\det H}}$$
(4.6)

where

$$g_0(\Delta) = \int_0^\infty dv \, g(v^2 + \Delta) \, v^{n-1}$$
(4.7)

The generating function is given by (2.4). Following the same procedure as above one obtains

$$\mathcal{G}(\mathbf{k}) = \int d^{n} \mathbf{y} \,\rho(\mathbf{y} + \mathbf{h}) \,e^{-i\mathbf{k} \cdot (\mathbf{y} + \mathbf{h})} \tag{4.8a}$$

$$= \frac{N e^{-i\mathbf{k} \cdot \mathbf{h}}}{\sqrt{\det H}} \int d^n \mathbf{v} \, g(v^2 + \Delta) \, e^{-i\mathbf{q} \cdot \mathbf{v}} \tag{4.8b}$$

$$= \frac{\Gamma(n/2) e^{-i\mathbf{k}\cdot\mathbf{h}}}{g_0(\Delta)} \int_0^\infty dv \, g(v^2 + \Delta) v^{n-1} \left(\frac{2}{qv}\right)^\nu J_\nu(qv) \,, \quad \nu = n/2 - 1 \tag{4.8c}$$

where $\mathbf{q} = H_D^{-1/2} U \mathbf{k}$. If $g(s) \to 0$ as $s \to \infty$ faster than any power of s we can Taylor-expand the Bessel function above and integrate term by term to obtain

$$\mathcal{G}(\mathbf{k}) = e^{-i\mathbf{k}\cdot\mathbf{h}} \sum_{\ell=0}^{\infty} A(n;\ell) \frac{(-q^2/2)^{\ell}}{\ell!}$$
(4.9)

where $A(n; \ell)$ is formally identical to Eq. (3.11) except that g_{ℓ} is now given by

$$g_{\ell}(\Delta) = \int_{0}^{\infty} dv \, g(v^{2} + \Delta) \, v^{n+2\ell-1}$$
(4.10)

rather than by (3.13).

The exponential term $e^{-i\mathbf{k}\cdot\mathbf{h}}$ in Eq. (4.9) leads to nonvanishing odd-order moments. An easy way to compute the moments is obtained by setting $\mathbf{x} = \mathbf{y} + \mathbf{h}$ so that, from Eq. (4.8a), one gets

$$\langle x_i x_j \cdots \rangle = \langle (y_i + h_i)(y_j + h_j) \cdots \rangle \tag{4.11}$$

Now Eq. (4.8a) is suggestively written in the form

$$\mathcal{G}(\mathbf{k}) = e^{-i\mathbf{k}\cdot\mathbf{h}} \mathcal{G}_y(\mathbf{k}) \tag{4.12}$$

where $\mathcal{G}_{y}(\mathbf{k})$ is the generating function for the *y*-moments. Eq. (4.9) implies that

$$\mathcal{G}_{y}(\mathbf{k}) = \int d^{n} \mathbf{y} \,\rho(\mathbf{y} + \mathbf{h}) \,e^{-i\mathbf{k} \cdot \mathbf{y}} = \sum_{\ell=0}^{\infty} A(n;\ell) \frac{(-q^{2}/2)^{\ell}}{\ell!} \tag{4.13}$$

which, in turn, implies that the even-order moments for the y's are given by (3.14). For example,

$$\langle x_i \rangle = \langle y_i + h_i \rangle = h_i \tag{4.14a}$$

$$\langle x_i x_j \rangle = \langle (y_i + h_i)(y_j + h_j) \rangle = \langle y_i y_j \rangle + h_i h_j = (g_1/ng_0)G_{ij} + h_i h_j$$
(4.14b)

$$\langle x_i x_j x_k \rangle = \langle (y_i + h_i)(y_j + h_j)(y_k + h_k) \rangle = \langle y_i y_j \rangle h_k + \langle y_j y_k \rangle h_i + \langle y_k y_i \rangle h_j + h_i h_j h_k$$

$$= (g_1/ng_0) (G_{ij}h_k + G_{jk}h_i + G_{ki}h_j) + h_i h_j h_k$$
(4.14c)

etc.

5 Subspace Projection.

Suppose now that the arbitrary function $f(\mathbf{x})$ depends only on a subset of components of \mathbf{x} , namely $f = f(\mathbf{x}')$ where \mathbf{x}' is the *m*-dimensional vector $\mathbf{x}' = (x_1, x_2, \dots, x_m)$ such that $1 \leq m \leq n-1$. Then a projected density $\rho'(\mathbf{x}')$ can be defined such that

$$\langle f(\mathbf{x}')\rangle = \int d^m \mathbf{x}' \rho'(\mathbf{x}') f(\mathbf{x}')$$
 (5.1)

This implies that

$$\rho'(\mathbf{x}') = \int d^{n-m} \mathbf{y} \,\rho(\mathbf{x}', \mathbf{y}) \tag{5.2}$$

where $\mathbf{y} = (x_{m+1}, x_{m+2}, \cdots, x_n).$

For the general quadratic density, $\rho(\mathbf{x}) = Ng(\langle \mathbf{x}|H|\mathbf{x} \rangle + \langle \mathbf{x}|\mathbf{a} \rangle)$, we start by partitioning H, \mathbf{x} and \mathbf{a} in the form

$$H = \begin{pmatrix} S & E \\ \hline F & K \end{pmatrix}, \quad \mathbf{x} = \begin{pmatrix} \mathbf{x}' \\ \hline \mathbf{y} \end{pmatrix}, \quad \mathbf{a} = \begin{pmatrix} \mathbf{c} \\ \hline \mathbf{d} \end{pmatrix}$$
(5.3)

where the vectors **c** and **d** are of dimension m and n-m, respectively, and where S is $m \times m$, E is $m \times (n-m)$, F is $(n-m) \times m$ and K is $(n-m) \times (n-m)$, with $S^T = S$, $K^T = K$ and $E^T = F$. This partition leads to

$$\langle \mathbf{x}|H|\mathbf{x}\rangle + \langle \mathbf{x}|\mathbf{a}\rangle = \langle \mathbf{y}|K|\mathbf{y}\rangle + 2\langle \mathbf{y}|F|\mathbf{x}'\rangle + \langle \mathbf{y}|\mathbf{d}\rangle + \langle \mathbf{x}'|S|\mathbf{x}'\rangle + \langle \mathbf{x}'|\mathbf{c}\rangle$$
(5.4)

Next we shift the origin of \mathbf{y} by setting $\mathbf{y} = \mathbf{y}' + \mathbf{h}$, substitute this in Eq. (5.4), and choose \mathbf{h} such that the term linear in \mathbf{y}' vanishes. A straightforward examination of the resultant expression shows that that the choice $\mathbf{h} = -K^{-1}(E\mathbf{x}' + \mathbf{d}/2)$ accomplishes the desired cancellation, hence

$$\langle \mathbf{x}|H|\mathbf{x}\rangle + \langle \mathbf{x}|\mathbf{a}\rangle = \langle \mathbf{y}'|K|\mathbf{y}'\rangle + \Delta'(\mathbf{x}')$$
(5.5)

where

$$\Delta'(\mathbf{x}') = \langle \mathbf{x}' | H' | \mathbf{x}' \rangle - \langle \mathbf{x}' | EK^{-1} | \mathbf{d} \rangle + \langle \mathbf{x}' | \mathbf{c} \rangle - \frac{1}{4} \langle \mathbf{d} | K^{-1} | \mathbf{d} \rangle$$
(5.6)

where we have defined

Z

$$H' \equiv S - EK^{-1}F \tag{5.7}$$

In the above formulas, K^{-1} is guaranteed to exists on account of Corollary 1, which implies that K is a symmetric positive-definite matrix. For this reason, we can set $\mathbf{y}' = V^{-1}K_D^{-1/2}\mathbf{v}$ where V is the orthogonal matrix that diagonalizes K, namely $VKV^{-1} = K_D$. Since $V^T = V^{-1}$ we obtain $\langle \mathbf{y}' | K | \mathbf{y}' \rangle = \langle \mathbf{v} | K_D^{-1/2} V K V^{-1} K_D^{-1/2} | \mathbf{v} \rangle = v^2$ hence the projected density is

$$\rho'(\mathbf{x}') = \int d^{n-m} \mathbf{y}' \rho(\mathbf{x}', \mathbf{y}' + \mathbf{h})$$

$$= \frac{N}{\sqrt{\det K}} \int d^{n-m} \mathbf{v} \, g(v^2 + \Delta'(\mathbf{x}'))$$
$$= \frac{N\Omega_{n-m}}{\sqrt{\det K}} \int_0^\infty dv \, g(v^2 + \Delta'(\mathbf{x}')) v^{n-m-1}$$
(5.8)

where N is given by Eqs. (4.6-4.7), hence the full expression for the projected density is

$$\rho'(\mathbf{x}') = \frac{\Omega_{n-m}}{\Omega_n g_0(\Delta)} \sqrt{\frac{\det H}{\det K}} \int_0^\infty dv \, g(v^2 + \Delta'(\mathbf{x}')) v^{n-m-1}$$
(5.9)

where $g_0(\Delta)$ is given by Eq. (4.7).

In Fourier space, subspace projection is straightforward. If $\tilde{\rho}(\mathbf{k})$ if the Fourier transform of $\rho(\mathbf{x})$, then the Fourier transform $\tilde{\rho}'(\mathbf{k}')$ of $\rho'(\mathbf{x}')$ is simply

$$\tilde{\rho}'(\mathbf{k}') = \tilde{\rho}(k_1, k_2, \cdots, k_m, 0, 0, \cdots, 0)$$
(5.10)

or, equivalently, the generating function $\mathcal{G}'(\mathbf{k}')$ is simply given by

$$\mathcal{G}'(\mathbf{k}') = \mathcal{G}(k_1, k_2, \cdots, k_m, 0, 0, \cdots, 0)$$
(5.11)

where \mathbf{k}' is the vector $\mathbf{k}' = (k_1, k_2, \cdots, k_m)$. As a result,

$$\langle f(\mathbf{x}')\rangle = \int \frac{d^m \mathbf{k}'}{(2\pi)^m} \left\langle e^{i\mathbf{k}'\cdot\mathbf{x}'} \right\rangle \tilde{f}(\mathbf{k}') = \int \frac{d^m \mathbf{k}'}{(2\pi)^m} \mathcal{G}'(-\mathbf{k}')\tilde{f}(\mathbf{k}')$$
(5.12)

This result confirms the obvious fact, which is true by the construction of $\rho'(\mathbf{x}')$, that the moments $C_{ij\cdots}^{(r)'}$, derived from $\rho'(\mathbf{x}')$, are exactly the same as the $C_{ij\cdots}^{(r)}$ derived from $\rho(\mathbf{x})$ provided the indices i, j, \cdots are constrained to the range $1 \leq i, j, \cdots \leq m$. In other words, for the specific case of r = 2, we can state that the correlation matrix $C^{(2)'}$ is the $m \times m$ upper-left submatrix of $C^{(2)}$.

Consider now the case of a strictly quadratic action, namely $\mathbf{a} = 0$. In this case $\Delta = 0$ and Eq. (5.9) yields

$$\rho'(\mathbf{x}') = \frac{\Gamma(n/2)}{g_0 \pi^{m/2} \Gamma((n-m)/2)} \sqrt{\frac{\det H}{\det K}} \int_0^\infty dv \, g(v^2 + s'(\mathbf{x}')) v^{n-m-1}$$
(5.13)

where g_0 is given by Eq. (3.6) and $s'(\mathbf{x}')$ is the "reduced action," given by

$$s'(\mathbf{x}') = \langle \mathbf{x}' | (S - EK^{-1}F) | \mathbf{x}' \rangle$$
(5.14)

Defining $\rho'(\mathbf{x}') = N'g'(s'(\mathbf{x}'))$ we obtain, up to an irrelevant multiplicative ambiguity,

$$g'(s'(\mathbf{x}')) = \int_{0}^{\infty} dv \, g(v^2 + s'(\mathbf{x}'))v^{n-m-1}$$
(5.15)

We can now apply the general result (3.26) to compute the 2nd-order moment for the *m*-dimensional density (5.13) to obtain

$$\langle x'_i x'_j \rangle = C_{ij}^{\prime(2)} = \frac{g'_1}{mg'_0} G'_{ij}$$
(5.16)

where $G'^{-1} = H'$.

According to the general formula (3.13), the moment g'_{ℓ} obtained from (5.15) is

$$g'_{\ell} = \int_{0}^{\infty} du \, u^{m+2\ell-1} \int_{0}^{\infty} dv \, g(v^2 + u^2) v^{n-m-1}$$
(5.17)

where we made the change of variables $s' = u^2$. Using now the change of variables $(u, v) = (t \cos \phi, t \sin \phi)$ where $0 \le t < \infty$ and $0 \le \phi \le \pi/2$ yields

$$g'_{\ell} = \frac{g_{\ell}}{2} B\left(\frac{n-m}{2}, \frac{m+2\ell}{2}\right)$$
(5.18)

where $B(\mu,\nu)$ is Euler's beta function. This result implies $g'_1/g'_0 = mg_1/ng_0$ so that Eq. (5.16) yields

$$C'^{(2)} = \frac{g_1}{ng_0}G' = \frac{g_1}{ng_0}H'^{-1}$$
(5.19)

Since $\rho'(\mathbf{x}')$ is normalized to unity we conclude, following the same integration procedure applied to Eq. (5.13), that

$$\det H' = \det(S - EK^{-1}F) = \frac{\det H}{\det K}$$
(5.20)

To prove this, we integrate Eq. (5.13) to obtain

$$\int d^{m} \mathbf{x}' \rho'(\mathbf{x}') = \frac{\Omega_{n-m}}{\Omega_{n} g_{0}} \sqrt{\frac{\det H}{\det K}} \times \int_{0}^{\infty} dv \, v^{n-m-1} \int d^{m} \mathbf{x}' g(v^{2} + \Delta'(\mathbf{x}'))$$
$$= \frac{\Omega_{n-m} \Omega_{m} g'_{0}}{\Omega_{n} g_{0}} \sqrt{\frac{\det H}{\det K \det H'}}$$
(5.21)

where

$$g_0' = \int_0^\infty du \, u^{m-1} \int_0^\infty dv \, g(v^2 + u^2) v^{n-m-1}$$

= $\frac{g_0}{2} B\left(\frac{n-m}{2}, \frac{m}{2}\right)$ (5.22)

Substituting the definition of $B(\mu, \nu)$ into (5.21) we obtain

$$1 = \sqrt{\frac{\det H}{\det K \det H'}} \tag{5.23}$$

which completes the proof.

Comparing Eq. (5.19) with $C^{(2)} = (g_1/ng_0)H^{-1}$ we arrive at the following theorem:

Theorem 5.1 Consider a real, symmetric, positive-definite, $n \times n$ matrix G whose inverse is $G^{-1} = H$. Consider the upper-left $m \times m$ submatrix G' of G, with $1 \le m \le n-1$. Then $G'^{-1} = S - EK^{-1}F$, where S, E, F and K are the matrices that result from partitioning H in the form shown in Eq. (5.3). Furthermore, $\det(S - EK^{-1}F) = \det H/\det K$.

Corollary 3: The above theorem, combined with Corollary 2, implies that $S - EK^{-1}F$ is a positive-definite matrix since G' is a positive-definite matrix.

A simplified proof of this theorem follows from computing the subspace projection of a Gaussian density of a strictly quadratic action, as shown in Sec. 6.1.1 below.

6 Examples.

In this section we provide several explicit examples of strictly-quadratic densities, and compute the normalization N according to Eq. (3.5), their moments g_{ℓ} according to Eq. (3.13), and their generating function $\mathcal{G}(\mathbf{k})$ according to Eq. (3.8). Table 1 summarizes the results.

6.1 Gaussian Density.

In this case $g(s) = e^{-s/2}$ and the explicit form of $\rho(\mathbf{x})$ is

$$\rho(\mathbf{x}) = \frac{\sqrt{\det H}}{(2\pi)^{n/2}} \exp\left\{-\frac{1}{2}\langle \mathbf{x}|H|\mathbf{x}\rangle\right\}$$
(6.1)

from which it follows that

$$g_{\ell} = 2^{n/2 + \ell - 1} \Gamma(n/2 + \ell) \tag{6.2}$$

$$A(n;\ell) = 1 \tag{6.3}$$

The fact that the g_{ℓ} 's exist for all ℓ is a consequence of the fact that $g(s) \to 0$ as $s \to \infty$ faster than any power of s, hence all moments exist. The fact that $A(n; \ell) = 1$ for all n and ℓ implies that $B(n; \ell) = D(n; \ell) = 1$ hence we recover the well-known results for the recursion formulas for the higher-order moments, Eq. (1.1).

The generating function is

$$\mathcal{G}(\mathbf{k}) = \sum_{\ell=0}^{\infty} \frac{(-q^2/2)^{\ell}}{\ell!} = \exp(-q^2/2)$$
(6.4)

which can also be obtained from the familiar Gaussian integration formula

$$\mathcal{G}(\mathbf{k}) = \frac{\sqrt{\det H}}{(2\pi)^{n/2}} \int d^n \mathbf{x} \exp\left\{-\frac{1}{2} \langle \mathbf{x} | H | \mathbf{x} \rangle - i \langle \mathbf{k} | \mathbf{x} \rangle\right\} = \exp\left\{-\frac{1}{2} \langle \mathbf{k} | G | \mathbf{k} \rangle\right\}$$
(6.5)

The 2nd-order moment $\langle x_i x_j \rangle = H_{ij}^{-1}$ or, indeed, any even-order moment, can also be obtained by differentiating both sides of the equation

$$\int d^{n}\mathbf{x} \,\exp\{-\frac{1}{2}\langle\mathbf{x}|H|\mathbf{x}\rangle\} = \frac{(2\pi)^{n/2}}{\sqrt{\det H}}$$
(6.6)

with respect to H_{ij} and using the formula $\partial \det M / \partial M_{ij} = (\det M) M_{ji}^{-1}$, valid for any nonsingular matrix M.

Eq. (6.3) implies that

$$G = H^{-1} = C^{(2)} = \begin{pmatrix} \langle x_1^2 \rangle & \langle x_1 x_2 \rangle & \cdots & \langle x_1 x_n \rangle \\ \langle x_1 x_2 \rangle & \langle x_2^2 \rangle & \cdots & \langle x_2 x_n \rangle \\ \vdots & \vdots & \ddots & \vdots \\ \langle x_1 x_n \rangle & \langle x_2 x_n \rangle & \cdots & \langle x_n^2 \rangle \end{pmatrix}$$
(6.7)

6.1.1 Subspace Projection for the Gaussian Density.

Using Eqs. (5.13-5.14) we obtain

$$\rho'(\mathbf{x}') = \frac{1}{(2\pi)^{m/2}} \sqrt{\frac{\det H}{\det K}} \exp\left\{-\frac{1}{2} \langle \mathbf{x}' | H' | \mathbf{x}' \rangle\right\}$$
(6.8)

where $\mathbf{x}' = (x_1, x_2, \dots, x_m)$ and $H' = S - EK^{-1}F$. Here the matrices S, E, F and K are defined by the partition of H shown in Eq. (5.3). This result states that the dimensionally-projected Gaussian density is

also Gaussian. This unique property of the Gaussian density is a direct consequence of the composition rule $\exp(x) \exp(y) = \exp(x+y)$ of the exponential function.

Since $\rho'(\mathbf{x}')$ is normalized to unity we conclude, comparing Eq. (6.8) with (6.1), that

$$\det H' = \frac{\det H}{\det K} \tag{6.9}$$

in accordance with the general Theorem 5.1.

The generating function for the dimensionally-projected Gaussian density is

$$\mathcal{G}'(\mathbf{k}') = \exp\left\{-\frac{1}{2}\langle \mathbf{k}' | G' | \mathbf{k}' \rangle\right\}$$
(6.10)

where $\mathbf{k}' = (k_1, k_2, \dots, k_m)$ and G' is the $m \times m$ upper-left submatrix of G, obtained from G by eliminating all rows and all columns higher than m.

6.2 Gamma-Function Density.

This is a slight generalization of the Gaussian density, defined by $g(s) = s^{\alpha} e^{-s/2}$, for which all moments exist provided α is large enough, as explained below. We find

$$g_{\ell} = 2^{n/2 + \alpha + \ell - 1} \Gamma(n/2 + \alpha + \ell) \tag{6.11}$$

$$A(n;\ell) = \frac{(n/2 + \alpha)_{\ell}}{(n/2)_{\ell}}$$
(6.12)

$$\mathcal{G}(\mathbf{k}) = {}_{1}F_{1}(n/2 + \alpha; n/2; -q^{2}/2)$$
(6.13)

where ${}_{1}F_{1}$ is the confluent hypergeometric function [10]. Clearly, if α is sufficiently negative the density will not be normalizable as a consequence of the divergence of $\rho(\mathbf{x})$ in the neighborhood of $\mathbf{x} = 0$. The normalization exists (*i.e.*, g_{0} is finite) if $\alpha > -n/2$. This condition ensures the existence of all the g_{ℓ} 's.

6.3 Exponential Density.

This case is defined by the choice $g(s) = e^{-\sqrt{s}}$; defining $\nu = (n+1)/2$ we obtain

$$g_{\ell} = \Gamma(n+2\ell) \tag{6.14}$$

$$A(n;\ell) = 2^{\ell}(\nu)_{\ell}$$
(6.15)

$$\mathcal{G}(\mathbf{k}) = (1+q^2)^{-\nu} \tag{6.16}$$

6.4 Generalized Lorentzian Density.

Here we choose $g(s) = (s+1)^{-p}$. This density extends over all space, but $\rho(\mathbf{x})$ falls off as $|\mathbf{x}|^{-2p}$ at infinity. Defining $\mu = p - n/2$ we obtain

$$g_{\ell} = \frac{\Gamma(n/2 + \ell)\Gamma(\mu - \ell)}{2\Gamma(p)}, \qquad \ell < \mu$$
(6.17)

$$A(n;\ell) = \frac{\Gamma(\mu-\ell)}{2^{\ell}\Gamma(\mu)}, \qquad \ell < \mu$$
(6.18)

(the g_{ℓ} 's and $A(n; \ell)$'s do not exist for $\ell \geq \mu$).

In this case it is not legitimate to obtain $\mathcal{G}(\mathbf{k})$ by term-by-term integration of the Taylor expansion of the Bessel function in Eq. (3.8). Instead, we have :

$$\mathcal{G}(\mathbf{k}) = \frac{2\Gamma(p)}{\Gamma(\mu)} \left(\frac{2}{q}\right)^{\nu} \int_{0}^{\infty} dv \, \frac{v^{n/2} J_{\nu}(qv)}{(v^2+1)^p}, \qquad \nu = n/2 - 1$$
(6.19a)

$$=\frac{2}{\Gamma(\mu)}\left(\frac{q}{2}\right)^{\mu}K_{\mu}(q) \tag{6.19b}$$

$$= \frac{1}{\Gamma(\mu)} \sum_{0 \le \ell < \mu} \Gamma(\mu - \ell) \frac{(-q^2/4)^{\ell}}{\ell!} + \text{infinite terms}$$
(6.19c)

where $K_{\mu}(z)$ is the usual modified Bessel function [10] of order $\mu = p - n/2$. For the distribution to exist, *i.e.*, to be normalizable, the condition p > n/2, or $\mu > 0$, must be satisfied. Eq. (6.18) shows that the moments exist only for values of ℓ such that $\ell < \mu$, or r < 2p - n, in consequence of the behavior $\rho(\mathbf{x}) \sim |\mathbf{x}|^{-2p}$ as $x \to \infty$. Although the generating function $\mathcal{G}(\mathbf{k})$, given by Eq. (6.19b), exists for all \mathbf{k} when $\mu > 0$, its Taylor expansion at $\mathbf{k} = 0$ does not formally exist because its *r*th derivative at $\mathbf{k} = 0$ is finite only as long as $r < 2\mu$, in full agreement with the discussion on densities with a power-law fall-off at infinity following Eq. (2.3). This is the meaning of the symbol "=" in Eq. (6.19c).

6.4.1 Subspace Projection for the Lorentzian Density.

For the strictly quadratic action $(\mathbf{a} = 0)$, Eq. (5.13) yields

$$\rho'(\mathbf{x}') = \frac{2\Gamma(p)\sqrt{\det H'}}{\pi^{m/2}\Gamma(\mu)\Gamma((n-m)/2)} \int_{0}^{\infty} dv \, \frac{v^{n-m-1}}{(v^2 + s'(\mathbf{x}'))^p} = \frac{\sqrt{\det H'}}{\pi^{m/2}\Gamma(\mu)} \frac{\Gamma(p')}{(s'(\mathbf{x}') + 1)^{p'}}$$
(6.20)

where p' = p - (n - m)/2 and where $s'(\mathbf{x}')$ is given by Eq. (5.14). The projected density is to be compared with the original,

$$\rho(\mathbf{x}) = \frac{\sqrt{\det H}}{\pi^{n/2} \Gamma(\mu)} \frac{\Gamma(p)}{(s(\mathbf{x}) + 1)^p}$$
(6.21)

which shows that the form is maintained (the projected density is also a Lorentzian), although the power p is shifted by (n-m)/2. Note that $\mu = p - n/2$ can be rewritten in the suggestive form $\mu = p' - m/2$, hence Eq. (6.20) can be obtained from (6.21) by the replacements $s(\mathbf{x}) \to s'(\mathbf{x}')$, det $H \to \det H'$, $n \to m$ and $p \to p'$, as it should be expected.

6.5 Microcanonical Density.

This case corresponds to the choice $g(s) = \delta(s-1)$ in the general formula (3.7). We call it "microcanonical" because it corresponds to the microcanonical distribution in statistical mechanics for quadratic Hamiltonians. It also corresponds to the "airbag," or KV, distribution in beam physics [11]. We obtain

$$g_{\ell} = 1/2$$
 (6.22)

$$A(n;\ell) = \frac{1}{2^{\ell} (n/2)_{\ell}}$$
(6.23)

$$\mathcal{G}(\mathbf{k}) = \Gamma(\nu+1) \left(\frac{2}{q}\right)^{\nu} J_{\nu}(q), \qquad \nu = n/2 - 1$$
(6.24)

6.5.1 Subspace Projection for the Microcanonical Density.

Setting $\mathbf{a} = 0$ in Eq. (5.13) yields, for this particular example,

$$\rho'(\mathbf{x}') = \frac{2\Gamma(n/2)\sqrt{\det H'}}{\pi^{m/2}\Gamma((n-m)/2)} \int_{0}^{\infty} dv \,\delta(v^2 + s'(\mathbf{x}') - 1)v^{n-m-1}$$
$$= \frac{\Gamma(n/2)\sqrt{\det H'}}{\pi^{m/2}\Gamma(\beta+1)} \left(1 - s'(\mathbf{x}')\right)^{\beta} \theta(1 - s'(\mathbf{x}'))$$
(6.25)

where $\beta = (n - m)/2 - 1$ and where $s'(\mathbf{x}')$ is given by Eq. (5.14). The projected density is a generalized waterbag density (see Sec. 6.6.2) of dimension m and power $\beta = (n - m)/2 - 1$.

6.6 Beta-Function Density.

By "beta-function density" we mean the choice $g(s) = s^{\alpha}(1-s)^{\beta}\theta(1-s)$. This density is of finite extent, as it vanishes, by definition, for s > 1, hence all moments exist provided α and β are appropriately restricted, as spelled out below. We obtain:

$$g_{\ell} = \frac{\Gamma(n/2 + \alpha + \ell)\Gamma(\beta + 1)}{2\Gamma(n/2 + \alpha + \beta + \ell + 1)}$$
(6.26)

$$A(n;\ell) = \frac{(n/2 + \alpha)_{\ell}}{2^{\ell}(n/2)_{\ell}(n/2 + \alpha + \beta + 1)_{\ell}}$$
(6.27)

$$\mathcal{G}(\mathbf{k}) = {}_{1}F_{2}(n/2 + \alpha; n/2, n/2 + \alpha + \beta + 1; -q^{2}/4)$$
(6.28)

where $\nu = n/2 - 1$ and ${}_{1}F_{2}$ is a generalized hypergeometric function (not to be confused with the ordinary hypergeometric function ${}_{2}F_{1}$) [10]. The normalizability condition of $\rho(\mathbf{x})$ demands $\alpha > -n/2$ and $\beta > -1$.

6.6.1 Waterbag Density.

This case corresponds to the choice $\alpha = \beta = 0$ in the beta-function density, *i.e.*, $g(s) = \theta(1 - s)$, so that $\rho(\mathbf{x})$ is uniform for $0 \le s \le 1$. We obtain from the general formulas above

$$g_{\ell} = \frac{1}{n+2\ell} \tag{6.29}$$

$$A(n;\ell) = \frac{1}{2^{\ell}(n/2+1)_{\ell}}$$
(6.30)

$$\mathcal{G}(\mathbf{k}) = \Gamma(\nu+1) \left(\frac{2}{q}\right)^{\nu} J_{\nu}(q), \qquad \nu = n/2$$
(6.31)

6.6.2 Generalized Waterbag Density.

This corresponds to setting $\alpha = 0$ in the beta-function density, *i.e.*, $g(s) = (1 - s)^{\beta} \theta(1 - s)$, and we obtain

$$g_{\ell} = \frac{\Gamma(\beta+1)\Gamma(n/2+\ell)}{2\Gamma(n/2+\ell+\beta+1)}$$
(6.32)

$$A(n;\ell) = \frac{1}{2^{\ell} (n/2 + \beta + 1)_{\ell}}$$
(6.33)

$$\mathcal{G}(\mathbf{k}) = \Gamma(\nu+1) \left(\frac{2}{q}\right)^{\nu} J_{\nu}(q), \qquad \nu = n/2 + \beta$$
(6.34)

The normalizability condition of $\rho(\mathbf{x})$ demands $\beta > -1$.

6.6.3 Power Density.

This corresponds to setting $\beta = 0$ in the beta-function density, *i.e.*, $g(s) = s^{\alpha}\theta(1-s)$ so that

$$g_{\ell} = \frac{1}{n+2\alpha+2\ell} \tag{6.35}$$

$$\mathcal{G}(\mathbf{k}) = (n+2\alpha) \sum_{\ell=0}^{\infty} \frac{(-q^2/4)^{\ell}}{(n+2\alpha+2\ell)(n/2)_{\ell} \,\ell!}$$
(6.36)

The normalizability condition of $\rho(\mathbf{x})$ demands $\alpha > -n/2$.

7 Extension to the Complex Space.

In this section we will consider an action of the form

$$s(\mathbf{z}) = \langle \mathbf{z} | H | \mathbf{z} \rangle \equiv \sum_{ij=1}^{n} H_{ij} \bar{z}_i z_j$$
(7.1)

where \mathbf{z} is a complex *n*-dimensional vector and H is an $n \times n$ positive-definite Hermitian matrix $(H^* = H^T)$, so that $s(\mathbf{z})$ is real. In this case the *n*-dimensional complex vector \mathbf{z} is $\mathbf{z} = (z_1, z_2, \dots, z_n)$, the bra $\langle \mathbf{z} |$ and ket $|\mathbf{z}\rangle$ are defined by $|\mathbf{z}\rangle = \operatorname{col}(z_1, z_2, \dots, z_n)$ and $\langle \mathbf{z} | = (\bar{z}_1, \bar{z}_2, \dots, \bar{z}_n)$ (we use interchangeably the notations z^* and \bar{z} to denote complex conjugation). Although we could add a linear term to the action, of the form $\langle \mathbf{a} | \mathbf{z} \rangle + \langle \mathbf{z} | \mathbf{a} \rangle$, we shall restrict the discussion to strictly quadratic action of the form (7.1).

A real density is defined by

$$\rho(\mathbf{z}) = Ng(s(\mathbf{z})) \tag{7.2}$$

where g(s) is a real non-negative function of the real variable s. It is straightforward to prove that Theorems 3.1 and 5.1, along with Corollaries 1, 2 and 3 carry over to the complex case with the replacement "real symmetric matrix" \rightarrow "complex Hermitian matrix" and "orthogonal matrix" \rightarrow "unitary matrix." In particular, we conclude that $s(\mathbf{z}) \geq 0$, the equality holding if and only if $\mathbf{z} = 0$.

The normalization constant N, as well as any statistical average of a function of \mathbf{z} , is determined by an *n*-fold surface integral, one for each of the components z_i of \mathbf{z} , *i.e.*

$$N^{-1} = \int d^{2n} \mathbf{z} \, g(s(\mathbf{z})) \tag{7.3}$$

where

$$d^{2n}\mathbf{z} = \prod_{i=1}^{n} dx_i dy_i \tag{7.4}$$

where x_i and y_i are the real and imaginary parts of z_i .

The most straightforward way to extend the results from the real to the complex case is by casting the latter in terms of a real $2n \times 2n$ matrix. This is accomplished by splitting H and \mathbf{z} into their real and imaginary parts, H = A + iB and $\mathbf{z} = \mathbf{x} + i\mathbf{y}$, where A, B, \mathbf{x} and \mathbf{y} are real, so that

$$A = \frac{1}{2}(H + H^*) = A^T$$
(7.5a)

$$B = \frac{1}{2i}(H - H^*) = -B^T$$
(7.5b)

hence

$$s(\mathbf{z}) = \sum_{i,j=1}^{n} (A+iB)_{ij} (x_i - iy_i) (x_j + iy_j)$$

=
$$\sum_{i,j=1}^{n} (A_{ij} (x_i x_j + y_i y_j) - B_{ij} (x_i y_j - x_j y_i))$$
(7.6)

This expression can be cast in the form $s(\mathbf{z}) = \mathbf{X}^T \cdot M\mathbf{X} = \langle \mathbf{X} | M | \mathbf{X} \rangle$ where M is the real symmetric $2n \times 2n$ matrix

$$M = \begin{pmatrix} A & -B \\ B & A \end{pmatrix}$$
(7.7)

and \mathbf{X} is the real 2n-dimensional vector

$$\mathbf{X} = \begin{pmatrix} \mathbf{x} \\ \mathbf{y} \end{pmatrix} \tag{7.8}$$

It is now straightforward to prove that the eigenvalues of H are the same as those of M: let \mathbf{z} be an eigenvector of H with eigenvalue h, that is $H\mathbf{z} = h\mathbf{z}$ or

$$(A+iB)(\mathbf{x}+i\mathbf{y}) = h(\mathbf{x}+i\mathbf{y}) \tag{7.9}$$

Separating the real and imaginary parts and using the fact that h is real, this equation is written in the form $M\mathbf{X} = h\mathbf{X}$ which shows that h is an eigenvalue of M. Now this equation, combined with Eq. (7.7), implies that $M\mathbf{\tilde{X}} = h\mathbf{\tilde{X}}$ where $\mathbf{\tilde{X}} = \operatorname{col}(\mathbf{y}, -\mathbf{x})$, *i.e.* the vector $\mathbf{\tilde{X}}$ is also an eigenvector of M with eigenvalue h. Since $\mathbf{\tilde{X}}$ is orthogonal to (and therefore linearly independent of) \mathbf{X} , we conclude that each eigenvalue h of H is a doubly degenerate eigenvalue of M. A corollary of this result is that, if H is positive-definite, so is M. Another corollary is that det $M = (\det H)^2$. With these observations, the results of all previous sections carry over to this case with the replacements $n \to 2n$, $H \to M$ and det $H \to (\det H)^2$. All moments are expressed in terms of the matrix elements of M^{-1} , which is given, in terms of A and B, by

$$M^{-1} = \begin{pmatrix} (A + BA^{-1}B)^{-1} & (B + AB^{-1}A)^{-1} \\ -(B + AB^{-1}A)^{-1} & (A + BA^{-1}B)^{-1} \end{pmatrix} = \begin{pmatrix} \operatorname{Re}(H^{-1}) & -\operatorname{Im}(H^{-1}) \\ \operatorname{Im}(H^{-1}) & \operatorname{Re}(H^{-1}) \end{pmatrix}$$
(7.10)

where we have used the decomposition of H^{-1} into its real and imaginary parts,

$$H^{-1} = (A + iB)^{-1} = (A + BA^{-1}B)^{-1} - i(B + AB^{-1}A)^{-1}$$
(7.11)

This result implies that $\operatorname{Re}(H^{-1})$ is symmetric while $\operatorname{Im}(H^{-1})$ is antisymmetric, as it should be the case.

As an example, consider a Gaussian density $\rho(\mathbf{z}) = N \exp\{-\langle \mathbf{z} | H | \mathbf{z} \rangle\} = N \exp\{-\langle \mathbf{X} | M | \mathbf{X} \rangle\}$. The 2nd-order moments are given by the results of Sec. 6.1,

$$\langle x_i x_j \rangle = \langle y_i y_j \rangle = \frac{1}{2} (A + BA^{-1}B)_{ij}^{-1} = \frac{1}{2} \operatorname{Re}(H^{-1})_{ij}$$
 (7.12a)

$$\langle x_i y_j \rangle = -\langle y_i x_j \rangle = \frac{1}{2} (B + AB^{-1}A)_{ij}^{-1} = -\frac{1}{2} \text{Im}(H^{-1})_{ij}$$
 (7.12b)

where the factors 1/2 arise from the absence of the factor 1/2 in the exponent of $\exp\{-\langle \mathbf{X}|M|\mathbf{X}\rangle\}$ (cf. compare with Eq. (6.1)). In terms of the z's, this implies

$$\langle z_i z_j \rangle = \langle (x_i + iy_i)(x_j + iy_j) \rangle = 0$$
(7.13a)

$$\langle z_i \bar{z}_j \rangle = \langle (x_i + iy_i)(x_j - iy_j) \rangle = H_{ij}^{-1}$$
(7.13b)

As an equivalent alternative to the real 2*n*-dimensional space of the **x**'s and **y**'s, one may deal with the action (7.1) directly in the *n*-dimensional complex space of the **z**'s. For example, we compute the normalization constant N, given by

$$N^{-1} = \int d^{2n} \mathbf{z} \, g(\langle \mathbf{z} | H | \mathbf{z} \rangle) \tag{7.14}$$

by noting that H, being Hermitian, is diagonalized by a unitary matrix U, namely $H = U^{-1}H_DU$ where $H_D = \text{diag}(h_1, h_2, \dots, h_n)$ and $U^{-1} = U^{\dagger}$. Since $h_i > 0$ for all i, the matrix $H_D^{1/2}$ exists unambiguously hence we can define the change of variables $|\mathbf{v}\rangle = H_D^{1/2}U|\mathbf{z}\rangle$, or $\langle \mathbf{v}| = \langle \mathbf{z}|U^{\dagger}H_D^{1/2}$, yielding $s(\mathbf{z}) = \langle \mathbf{z}|H|\mathbf{z}\rangle = \langle \mathbf{v}|\mathbf{v}\rangle = v^2$ hence

$$N^{-1} = \frac{1}{\det H} \int d^{2n} \mathbf{v} \, g(v^2) = \frac{\Omega_{2n} \, g_0}{\det H} = \frac{2\pi^n g_0}{\Gamma(n) \det H}$$
(7.15)

where now

$$g_0 = \int_0^\infty dv \, g(v^2) v^{2n-1} = \frac{1}{2} \int_0^\infty ds \, g(s) s^{n-1}$$
(7.16)

The moment of the z_i 's and \bar{z}_i 's is defined by

$$C_{ij\cdots,km\cdots}^{(\ell_1,\ell_2)} \equiv \langle \underbrace{z_i z_j \cdots \bar{z}_k \bar{z}_m \cdots}_{\ell_1 \ z's} \rangle = \int d^{2n} \mathbf{z} \, \rho(\mathbf{z}) (\underbrace{z_i z_j \cdots \bar{z}_k \bar{z}_m \cdots}_{\ell_1 \ z's})$$
(7.17)

This expression implies that, if $\rho(\mathbf{z})$ is of finite extent, or if it falls off as $|\mathbf{z}| \to \infty$ faster than any power of z, then the moments exist for any ℓ_1 , ℓ_2 . If, on the other hand, $\rho(\mathbf{z}) \sim |\mathbf{z}|^{-2p}$ as $|\mathbf{z}| \to \infty$, then the moments exist only if $\ell_1 + \ell_2 < 2(p-n)$. Eq. (7.17) also implies that the moment $C_{ij\cdots,km\cdots}^{(\ell_1,\ell_2)}$ is fully symmetric under any permutation of the first group of indices (i, j, \cdots) and/or the second (k, m, \cdots) . Furthermore, these sets of indices are exchanged under complex conjugation, namely

$$C_{ij\dots,km\dots}^{(\ell_1,\ell_2)*} = C_{km\dots,ij\dots}^{(\ell_2,\ell_1)}$$
(7.18)

The generating function for the moments is defined by

$$\mathcal{G}(\mathbf{k}, \mathbf{q}) = \int d^{2n} \mathbf{z} \,\rho(\mathbf{z}) e^{-i(\langle \mathbf{k} | \mathbf{z} \rangle + \langle \mathbf{z} | \mathbf{q} \rangle)} \tag{7.19}$$

where \mathbf{k} and \mathbf{q} are two real,³ independent, *n*-dimensional vectors, so that

$$C_{ij\cdots,km\cdots}^{(\ell_1,\ell_2)} = i^{\ell_1+\ell_2} \underbrace{\frac{\partial^{\ell_1+\ell_2} \mathcal{G}(\mathbf{k},\mathbf{q})}{\partial k_i \partial k_j \cdots \partial q_k \partial q_m \cdots}}_{\ell_1 \ k's} \Big|_{\mathbf{k}=\mathbf{q}=0}$$
(7.20)

In case that $\rho(\mathbf{z})$ is of finite extent, or if it falls off as $|\mathbf{z}| \to \infty$ faster than any power of $|\mathbf{z}|$, then all moments exist and we can Taylor-expand the exponential factor in (7.19) about $\mathbf{k} = \mathbf{q} = 0$ to obtain

$$\mathcal{G}(\mathbf{k}, \mathbf{q}) = 1 - i \sum_{i=1}^{n} \left(k_i \langle z_i \rangle + q_i \langle \bar{z}_i \rangle \right) + \frac{(-i)^2}{2!} \sum_{i,j=1}^{n} \left(k_i k_j \langle z_i z_j \rangle + q_i q_j \langle \bar{z}_i \bar{z}_j \rangle + 2k_i q_j \langle z_i \bar{z}_j \rangle \right) + \cdots$$
(7.21a)
$$= 1 - i \sum_{i=1}^{n} \left(k_i C_{i,}^{(1,0)} + q_i C_{,i}^{(0,1)} \right) + \frac{(-i)^2}{2!} \sum_{i,j=1}^{n} \left(k_i k_j C_{ij,}^{(2,0)} + q_i q_j C_{,ij}^{(0,2)} + 2k_i q_j C_{i,j}^{(1,1)} \right) + \cdots$$
(7.21b)

In what remains of this section we confine our attention to strictly quadratic actions, of the form (7.1). In this case we note that the moment $C^{(\ell_1,\ell_2)}$ vanishes if $\ell_1 \neq \ell_2$. To prove this assertion, we make the change of variable $z_i \rightarrow z_i e^{i\alpha}$ in the right-hand side of (7.17), where α is a real constant. Since the integral over \mathbf{z} extends over all space, and the Jacobian of this transformation is unity, and $s(\mathbf{z})$ is invariant under this transformation, we obtain the identity

$$C_{ij\dots,km\dots}^{(\ell_1,\ell_2)} = e^{i(\ell_1 - \ell_2)\alpha} C_{ij\dots,km\dots}^{(\ell_1,\ell_2)}$$
(7.22)

 $^{^{3}}$ The analytic extension to complex **k** and **q** is straightforward but not necessary for our discussion.

which implies $C^{(\ell_1,\ell_2)} = 0$ if $\ell_1 \neq \ell_2$ in consequence of the arbitrariness of α . Using now the positivity of s one can write, in complete analogy with Eq. (3.7),

$$\rho(\mathbf{z}) = N \int_{0}^{\infty} ds \, g(s) \, \delta(\langle \mathbf{z} | H | \mathbf{z} \rangle - s) \tag{7.23}$$

hence

$$\mathcal{G}(\mathbf{k}, \mathbf{q}) = N \int_{0}^{\infty} ds \, g(s) \int d^{2n} \mathbf{z} \, \delta(\langle \mathbf{z} | H | \mathbf{z} \rangle - s) e^{-i(\langle \mathbf{k} | \mathbf{z} \rangle + \langle \mathbf{z} | \mathbf{q} \rangle)}$$
$$= \frac{\Gamma(n)}{g_0 u^{n-1}} \int_{0}^{\infty} dv \, g(v^2) v^n J_{n-1}(2vu)$$
(7.24)

where we have used Eq. (B.10), made the change of variable $s = v^2$, and defined $u = \sqrt{\langle \mathbf{k} | G | \mathbf{q} \rangle}$ and $G = H^{-1}$. Although $\langle \mathbf{k} | G | \mathbf{q} \rangle$ is not, in general, positive, no ambiguity arises in the definition of its square root since only even powers of u appear in Eq. (7.24).

If g(s) falls off as $s \to \infty$ faster than any power of s, we can Taylor-expand the Bessel function in Eq. (7.24) and integrate term by term yielding

$$\mathcal{G}(\mathbf{k},\mathbf{q}) = \sum_{\ell=0}^{\infty} A(n;\ell) \frac{(-u^2)^{\ell}}{\ell!}$$
(7.25)

where $A(n; \ell)$ is defined by

$$A(n;\ell) = \frac{g_\ell}{g_0(n)_\ell} \tag{7.26}$$

and g_{ℓ} is given by

$$g_{\ell} = \int_{0}^{\infty} dv \, g(v^2) v^{2(n+\ell)-1} = \frac{1}{2} \int_{0}^{\infty} ds \, g(s) s^{n+\ell-1}$$
(7.27)

In consequence of the vanishing of $C^{(\ell_1,\ell_2)}$ for $\ell_1 \neq \ell_2$, only the even terms contribute to the Taylor expansion (7.21), and of these, only those for which $\ell_1 = \ell_2$, thus

$$\mathcal{G}(\mathbf{k},\mathbf{q}) = 1 + \frac{(-i)^2}{2!} \binom{2}{1} \sum_{i,j=1}^n (k_i q_j) C_{i,j}^{(1,1)} + \frac{(-i)^4}{4!} \binom{4}{2} \sum_{i,j,k,m=1}^n (k_i k_j q_k q_m) C_{ij,km}^{(2,2)} + \cdots$$
(7.28a)

$$=\sum_{\ell=0}^{\infty} \frac{(-)^{\ell}}{\ell!^2} \sum_{\substack{i, j, \dots = 1 \\ k, m, \dots = 1}}^{n} (k_i k_j \cdots q_k q_m \cdots) C_{ij \cdots, km \cdots}^{(\ell, \ell)}$$
(7.28b)

By using the formula

$$\frac{\partial^{\ell}(x_{i_1}x_{i_2}\cdots x_{i_{\ell}})}{\partial x_{j_1}\partial x_{j_2}\cdots \partial x_{j_{\ell}}} = \delta_{i_1j_1}\delta_{i_2j_2}\cdots \delta_{i_{\ell}j_{\ell}} + \text{ permutations} \quad (\text{total of } 2\ell \text{ indices}) \tag{7.29}$$

where the x's represent here either the k's or the q's, and where the "permutations" are such as to make the right-hand side a fully symmetric tensor of rank 2ℓ (total of ℓ ! terms), one recovers the result (7.20) (the derivatives with respect to the k's in (7.20) will yield a factor ℓ !, as will the derivatives with respect to the q's; these two factors of ℓ ! will just cancel the ℓ !² in the denominator of (7.28b)). The moment $C_{ij\cdots,km\cdots}^{(\ell,\ell)}$ is obtained from Eqs. (7.20) and (7.25) by setting $\ell_1 = \ell_2$ and using the easily-proven result

$$\underbrace{\frac{\partial^{2\ell} \langle \mathbf{k} | G | \mathbf{q} \rangle^{\ell}}{\partial k_{i} \partial k_{j} \cdots \partial q_{k} \partial q_{m} \cdots}}_{\ell \ k' s} \underbrace{= \ell! S^{(\ell,\ell)}_{ij \cdots, km \cdots}}$$
(7.30)

where $S_{ij\cdots,km\cdots}^{(\ell,\ell)}$ is defined to be

$$S_{ij\cdots,km\cdots}^{(\ell,\ell)} = \left[\underbrace{G_{ik}G_{jm}\cdots}_{\ell \; G's} + \text{permutations}\right] \quad (\ell! \text{ terms, } 2\ell \text{ indices})$$
(7.31)

where the "permutations" are such as to make the right-hand side fully symmetric⁴ under any permutation of the first set of indices (i, j, \dots) and simultaneously under any permutation of the second set (k, m, \dots) . When this result is applied to either (7.24) or (7.25) we obtain the fundamental result

$$C_{ij\cdots,km\cdots}^{(\ell,\ell)} = \langle \underbrace{z_i z_j \cdots}_{\ell z' s} \overline{z_k \bar{z}_m \cdots}_{\ell \bar{z}' s} \rangle = A(n;\ell) S_{ij\cdots,km\cdots}^{(\ell,\ell)}$$
(7.32)

A few examples are:

$$C_{i,j}^{(1,1)} = \langle z_i \bar{z}_j \rangle = A(n;1)G_{ij}$$
 (7.33a)

$$C_{ij,km}^{(2,2)} = \langle z_i z_j \bar{z}_k \bar{z}_m \rangle = A(n;2) \left[G_{ik} G_{jm} + G_{im} G_{jk} \right]$$
(7.33b)

$$C_{ijk,\alpha\beta\gamma}^{(3,3)} = \langle z_i z_j z_k \bar{z}_\alpha \bar{z}_\beta \bar{z}_\gamma \rangle = A(n;3) \left[G_{i\alpha} G_{j\beta} G_{k\gamma} + \cdots \right] \qquad (6 \text{ terms})$$
(7.33c)

If $g(s) \sim s^{-p}$ as $s \to \infty$, corresponding to $\rho(\mathbf{z}) \sim |\mathbf{z}|^{-2p}$ as $\mathbf{z} \to \infty$, the moment $C_{ij\dots,km\dots}^{(\ell,\ell)}$ exists only if g_{ℓ} exists, and is given by expression (7.32), only if $\ell , in agreement with the discussion following Eq. (7.17).$

7.1 Recursion Formulas for the Moments.

Eq. (7.32) can be cast in the recursive form

$$C_{ijk\cdots,\alpha\beta\gamma\cdots}^{(\ell,\ell)} = B(n;\ell) \Big[C_{i,\alpha}^{(1,1)} C_{jk\cdots,\beta\gamma\cdots}^{(\ell-1,\ell-1)} + \text{permutations} \Big]$$
(7.34)

or, equivalently,

$$\langle z_i z_j \cdots \bar{z}_k \bar{z}_m \cdots \rangle = B(n; \ell) \Big[\langle z_i \bar{z}_k \rangle \langle z_j \cdots \bar{z}_m \cdots \rangle + \text{permutations} \Big]$$
(7.35)

where consistency demands the definition $C^{(0,0)} = 1$, and where

$$B(n;\ell) = \frac{A(n;\ell)}{A(n;\ell-1)A(n;1)} = \frac{ng_0g_\ell}{(n+\ell-1)g_1g_{\ell-1}}$$
(7.36)

From Eq. (7.32), the moment $C^{(\ell,\ell)}$ can also be expressed purely in terms of the $C^{(1,1)}$'s,

$$C_{ijk\cdots,\alpha\beta\gamma\cdots}^{(\ell,\ell)} = D(n;\ell) \left[\underbrace{C_{i,\alpha}^{(1,1)} C_{j,\beta}^{(1,1)} \cdots}_{\ell C^{(1,1)} \cdot s} + \text{permutations} \right]$$
(7.37)

or, equivalently,

$$\langle z_i z_j \cdots \bar{z}_k \bar{z}_m \cdots \rangle = D(n; \ell) \left[\underbrace{\langle z_i \bar{z}_k \rangle \langle z_j \bar{z}_m \rangle \cdots}_{\ell \text{ terms}} + \text{permutations} \right]$$
(7.38)

where

$$D(n;\ell) = \frac{A(n;\ell)}{A^{\ell}(n;1)} = \frac{g_{\ell}g_0^{\ell-1}}{g_1^{\ell}(n)_{\ell}}$$
(7.39)

⁴Note that G is Hermitian rather than symmetric, hence $G_{ji} = G_{ij}^*$.

7.2 Example: the Complex Gaussian Density.

For the case of the complex Gaussian density we choose $g(s) = e^{-s}$ (it is convenient not to include the factor 1/2 in the exponent in this complex case), thus

$$\rho(\mathbf{z}) = N \exp\{-\langle \mathbf{z} | H | \mathbf{z} \rangle\}$$
(7.40)

The normalization constant N is determined by Eqs. (7.15-7.16), yielding

$$N^{-1} = \int d^{2n} \mathbf{z} \, \exp\{-\langle \mathbf{z} | H | \mathbf{z} \rangle\} = \frac{\pi^n}{\det H}$$
(7.41)

The generating function follows from the integrals computed in Appendix B,

$$\mathcal{G}(\mathbf{k},\mathbf{q}) = N \int d^{2n} \mathbf{z} \, \exp\{-\langle \mathbf{z} | H | \mathbf{z} \rangle - i \langle \mathbf{k} | \mathbf{z} \rangle - i \langle \mathbf{z} | \mathbf{q} \rangle\} = \exp\{-\langle \mathbf{k} | H^{-1} | \mathbf{q} \rangle\}$$
(7.42)

and the g_{ℓ} 's are

$$g_{\ell} = \frac{1}{2} \int_{0}^{\infty} ds \, e^{-s} s^{n+\ell-1} = \frac{1}{2} \Gamma(n+\ell) \tag{7.43}$$

which implies $A(n; \ell) = 1$ for all n and ℓ , as expected from the comparison of Eqs. (7.25) and (7.42).

Taking $\partial/\partial k_i \partial q_j$ or $\partial/\partial k_i \partial k_j$ or $\partial/\partial q_i \partial q_j$ of (7.42), then setting $\mathbf{k} = \mathbf{q} = 0$ yields

$$\langle z_i \bar{z}_j \rangle = H_{ij}^{-1} \tag{7.44}$$

and $\langle z_i z_j \rangle = \langle \bar{z}_i \bar{z}_j \rangle = 0$, in agreement with Eqs. (7.13). The moment $\langle z_i \bar{z}_j \rangle$ or, indeed, any even-order moment, can also be obtained by differentiating both sides of Eq. (7.41) with respect to H_{ij} and using the formula $\partial \det M/\partial M_{ij} = (\det M)M_{ji}^{-1}$, valid for any nonsingular matrix M.

7.3 Subspace Projection.

The subspace projection of $\rho(\mathbf{z})$ is obtained in a similar fashion as in the real case, thus

$$\rho'(\mathbf{z}') = \int d^{2(n-m)} \mathbf{y} \,\rho(\mathbf{z}', \mathbf{y}) \tag{7.45}$$

where $\mathbf{y} = (z_{m+1}, z_{m+2}, \dots, z_n)$. For the strictly quadratic density, $\rho(\mathbf{z}) = Ng(\langle \mathbf{z}|H|\mathbf{z}\rangle)$, we partition H and \mathbf{z} in the form

$$H = \begin{pmatrix} S & E \\ \hline F & K \end{pmatrix}, \qquad \mathbf{z} = \begin{pmatrix} \mathbf{z}' \\ \hline \mathbf{y} \end{pmatrix}, \tag{7.46}$$

where S is $m \times m$, E is $m \times (n-m)$, F is $(n-m) \times m$ and K is $(n-m) \times (n-m)$, with $S^{\dagger} = S$, $K^{\dagger} = K$ and $E^{\dagger} = F$. This partition leads to

$$\langle \mathbf{z}|H|\mathbf{z}\rangle = \langle \mathbf{y}|K|\mathbf{y}\rangle + \langle \mathbf{y}|E|\mathbf{z}'\rangle + \langle \mathbf{z}'|F|\mathbf{y}\rangle + \langle \mathbf{z}'|S|\mathbf{z}'\rangle$$
(7.47)

By shifting **y** by a constant vector **h** in the form $\mathbf{y} = \mathbf{y}' + \mathbf{h}$ and choosing $|\mathbf{h}\rangle = -K^{-1}F|\mathbf{z}'\rangle$ or $\langle \mathbf{h}| = -\langle \mathbf{z}'|EK^{-1}$, we cancel the terms linear in \mathbf{y}' hence

$$\langle \mathbf{z}|H|\mathbf{z}\rangle = \langle \mathbf{y}'|K|\mathbf{y}'\rangle + s'(\mathbf{z}') \tag{7.48}$$

where we have defined the reduced action

$$s'(\mathbf{z}') = \langle \mathbf{z}' | (S - EK^{-1}F) | \mathbf{z}' \rangle \tag{7.49}$$

hence

$$\rho'(\mathbf{z}') = N \int d^{2(n-m)} \mathbf{y}' \, g(\langle \mathbf{y}' | K | \mathbf{y}' \rangle + s'(\mathbf{z}'))$$
(7.50)

In the above formulas, K^{-1} is guaranteed to exist on account of Corollary 1, which implies that K is a Hermitian positive-definite matrix. For this reason we can set $\mathbf{y}' = V^{-1}K_D^{-1/2}\mathbf{v}$ where V is the unitary matrix that diagonalizes K, namely $VKV^{-1} = K_D$. Using $V^{\dagger} = V^{-1}$ we obtain $\langle \mathbf{y}' | K | \mathbf{y}' \rangle = \langle \mathbf{v} | \mathbf{v} \rangle \equiv v^2$ hence the projected density is

$$\rho'(\mathbf{z}') = \frac{N}{\det K} \int d^{2(n-m)} \mathbf{v} \, g(v^2 + s'(\mathbf{z}')) = \frac{\Gamma(n) \det H}{g_0 \pi^m \Gamma(n-m) \det K} \int_0^\infty dv \, g(v^2 + s'(\mathbf{z}')) v^{2(n-m)-1}$$
(7.51)

where g_0 is given by Eq. (7.16). This result should be compared with (5.13).

8 Conclusions.

We have presented the computation of the moments and their generating functions for *n*-dimensional densities of the form $\rho(\mathbf{x}) \propto g(s(\mathbf{x}))$, where $s(\mathbf{x})$ is a positive quadratic form of the vector $\mathbf{x} = (x_1, x_2, \dots, x_n)$ and g(s) is a non-negative function of *s*. In particular, we have found the recursion relation that relates the *r*th-order moment with the (r-2)th-order moment. This result constitutes a generalization of the Gaussian moment theorem, which corresponds to the specific choice $g(s) = e^{-s/2}$. We have presented the extension of the above analysis to densities $\rho(\mathbf{z})$ defined over the *n*-dimensional complex space of vectors \mathbf{z} .

Acknowledgments.

I am grateful to A. Charman, A. Dragt and B. Shadwick for useful discussions.

References

- See, for example, C. W. Gardiner, Handbook of Stochastic Methods for Physics, Chemistry and the Natural Sciences (2nd ed.), Springer-Verlag, 1985.
- [2] G. C. Wick, "The Evaluation of the Collision Matrix," Phys. Rev. 80, 268 (1950).
- [3] J. E. Mayer and M. G. Mayer, Statistical Mechanics (2nd ed.), John Wiley & Sons, 1977, ch. 8.
- [4] A. J. Dragt, R. L. Gluckstern, F. Neri and G. Rangarajan, "Theory of Emittance Invariants," Proc. US-CERN School on Particle Accelerators, Capri, October 20–26, 1988, p. 94 (Springer-Verlag Lecture Notes in Physics 343; M. Month and S. Turner, eds.).
- [5] A. J. Dragt, F. Neri and G. Rangarajan, "General Moment Invariants for Linear Hamiltonian Systems," Phys. Rev. A 45, 2572 (1992).
- [6] P. A. M. Dirac, The Principles of Quantum Mechanics (4th ed.), Oxford, Clarendon Press, 1958.
- [7] M. Marcus and H. Minc, A Survey of Matrix Theory and Matrix Inequalities, Dover Publications, Inc., New York, 1992.
- [8] I. M. Gel'fand, Lectures on Linear Algebra, Dover Publications, Inc., New York.

- [9] Handbook of Mathematical Functions, M. Abramowitz and I. A. Stegun, eds., Dover Publications Inc., 9th printing, 1970.
- [10] N. N. Lebedev, Special Functions and Their Applications, Dover Publications, Inc., New York, 1972.
- [11] I. M. Kapchinskij and V. V. Vladimirskij, "Limitations of Proton Beam Current in a Strong Focusing Linear Accelerator Associated with the Beam Space Charge," Proc. Intl. Conf. on High Energy Accelerators and Instrumentation, 274–288 (1959) (CERN, Geneva, Sept. 14–19, 1959; L. Kowarski, ed.).
- [12] See, for example, Higher Transcendental Functions, A. Erdélyi et al., eds., vol. II of Bateman Manuscript Project, p. 232, McGraw-Hill, New York (1953).

A Certain *n*-Dimensional Integrals.

Consider the *n*-dimensional vector **x** with Cartesian coordinates (x_1, x_2, \dots, x_n) . The vector **x** can also be fully specified by its length $x = |\mathbf{x}|$ plus n - 1 angles $\theta_1, \theta_2, \dots, \theta_{n-1}$ such that [12]

$$\begin{cases} x_1 = x \cos \theta_1 \\ x_2 = x \sin \theta_1 \cos \theta_2 \\ x_3 = x \sin \theta_1 \sin \theta_2 \cos \theta_3 \\ \vdots \\ x_{n-1} = x \sin \theta_1 \sin \theta_2 \cdots \cos \theta_{n-1} \\ x_n = x \sin \theta_1 \sin \theta_2 \cdots \sin \theta_{n-1} \end{cases}$$
 where
$$\begin{cases} 0 \le \theta_i \le \pi, & i = 1, 2, \cdots, n-2 \\ 0 \le \theta_i \le 2\pi, & i = n-1 \end{cases}$$
(A.1)

The generalized coordinates $(q_i) = (x, \theta_1, \theta_2, \dots, \theta_{n-1})$ are orthogonal in the sense that

$$\frac{\partial \mathbf{x}}{\partial q_i} \cdot \frac{\partial \mathbf{x}}{\partial q_j} = 0 \qquad \text{for } i \neq j \tag{A.2}$$

The volume element is $d^n \mathbf{x} = x^{n-1} dx d\Omega_n$ where

$$d\Omega_n = (\sin\theta_1)^{n-2} (\sin\theta_2)^{n-3} \cdots (\sin\theta_{n-1})^0 d\theta_1 d\theta_2 \cdots d\theta_{n-1}$$
(A.3)

The surface of the unit sphere in n dimensions, Ω_n , is

$$\Omega_n = \int d^n \mathbf{x} \,\delta(|\mathbf{x}| - 1) = \int d\Omega_n$$

=
$$\int_0^{\pi} d\theta_1 \,(\sin\theta_1)^{n-2} \int_0^{\pi} d\theta_2 \,(\sin\theta_2)^{n-3} \cdots \int_0^{2\pi} d\theta_{n-1}$$

=
$$\frac{2\pi^{n/2}}{\Gamma(n/2)}$$
(A.4)

This result follows from repeated application of the formula

$$\int_{0}^{\pi} d\theta \left(\sin\theta\right)^{\nu} = \frac{\sqrt{\pi} \,\Gamma\left(\frac{\nu+1}{2}\right)}{\Gamma\left(\frac{\nu}{2}+1\right)} \tag{A.5}$$

which is valid for general complex ν provided only that $\operatorname{Re} \nu > -1$. A shortcut method to compute Ω_n is as follows: consider the integral

$$I_n = \int d^n \mathbf{x} \, e^{-\mathbf{x}^2} \tag{A.6}$$

By making use of the factorization property of the exponential function it is straightforward to carry out this integral in Cartesian coordinates, yielding $I_n = \pi^{n/2}$. On the other hand, going to *n*-dimensional spherical coordinates, we obtain

$$I_n = \int d\Omega_n \int_0^\infty dx \, x^{n-1} e^{-x^2} = \frac{1}{2} \Omega_n \int_0^\infty dt \, t^{n/2-1} e^{-t} = \frac{1}{2} \Omega_n \Gamma(n/2)$$
(A.7)

By equating the two results we obtain (A.4).

The volume of the unit sphere is

$$V_n = \int d^n \mathbf{x} \,\theta(1 - |\mathbf{x}|) = \int_0^1 dx \, x^{n-1} \int d\Omega_n = \frac{\Omega_n}{n} = \frac{\pi^{n/2}}{\Gamma(n/2 + 1)}$$
(A.8)

If a function $f(\mathbf{x})$ depends only on $|\mathbf{x}|$ and/or on a few components of \mathbf{x} , the integral $\int d^n \mathbf{x} f(\mathbf{x})$ can be reduced by integrating out the components on which $f(\mathbf{x})$ does not depend, *e.g.*,

$$\int d^{n}\mathbf{x} f(x) = \Omega_{n} \int_{0}^{\infty} dx \, x^{n-1} f(x)$$
(A.9)

$$\int d^{n}\mathbf{x} f(x, \mathbf{k} \cdot \mathbf{x}) = \Omega_{n-1} \int_{0}^{\infty} dx \, x^{n-1} \int_{0}^{\pi} d\theta_1 \, (\sin \theta_1)^{n-2} \, f(x, kx \cos \theta_1) \tag{A.10}$$

etc. (in this last integral we chose the "1" axis of \mathbf{x} aligned with the vector \mathbf{k}).

With these results, the computation of the integral found in Eq. (3.8),

$$I(\mathbf{k}) \equiv \int d^{n} \mathbf{x} \,\delta(\langle \mathbf{x} | H | \mathbf{x} \rangle - s) e^{-i\mathbf{k} \cdot \mathbf{x}}$$
(A.11)

proceeds as follows: we first make the change of variables $\mathbf{x} = U^{-1}H_D^{-1/2}U\mathbf{v}$ where U is a real orthogonal matrix that diagonalizes H, namely $H = U^{-1}H_DU$, hence

$$I(\mathbf{k}) = \frac{1}{\sqrt{\det H}} \int d^n \mathbf{v} \,\delta(v^2 - s) e^{-i\mathbf{q}\cdot\mathbf{v}} \tag{A.12}$$

where we used $\langle \mathbf{x}|H|\mathbf{x}\rangle = \mathbf{v}^2$, defined the vector $\mathbf{q} = H_D^{-1/2} U \mathbf{k}$, and where the prefactor $(\det H)^{-1/2}$ is the Jacobian of the transformation $\mathbf{x} \to \mathbf{v}$. This integral is now in the form of Eq. (A.10), hence

$$I(\mathbf{k}) = \frac{\Omega_{n-1}}{\sqrt{\det H}} \int_{0}^{\infty} dv \, v^{n-1} \delta(v^2 - s) \int_{0}^{\pi} d\theta_1 \, (\sin \theta_1)^{n-2} e^{-iqv \cos \theta_1} \tag{A.13a}$$

$$= \frac{\pi^{n/2}}{\sqrt{\det H}} \left(\frac{2\sqrt{s}}{q}\right)^{\nu} J_{\nu}(q\sqrt{s}) \tag{A.13b}$$

$$= \frac{\pi^{n/2} s^{\nu}}{\sqrt{\det H}} \sum_{\ell=0}^{\infty} \frac{(-q^2 s/4)^{\ell}}{\ell! \Gamma(n/2+\ell)}$$
(A.13c)

where $\nu = n/2 - 1$, $q = |\mathbf{q}| = \sqrt{\langle \mathbf{k} | H^{-1} | \mathbf{k} \rangle}$, and we used a standard definition [10] of the ordinary Bessel function $J_{\nu}(z)$.

B Complex Integrals.

In this section the area element d^2z for a surface integral over the complex plane of the variable z = x + iyis defined as $d^2z = dxdy$. The *n*-dimensional complex vector \mathbf{z} is $\mathbf{z} = (z_1, z_2, \dots, z_n)$, the ket $|\mathbf{z}\rangle$ and bra $\langle \mathbf{z}|$ are defined by $|\mathbf{z}\rangle = \operatorname{col}(z_1, z_2, \dots, z_n)$ and $\langle \mathbf{z}| = (\bar{z}_1, \bar{z}_2, \dots, \bar{z}_n)$, respectively, and H is a Hermitian positive-definite matrix. We denote complex conjugation by either a bar or an asterisk.

The two basic integrals are

$$\int d^2 z \, e^{-a|z|^2} = \frac{\pi}{a} \,, \quad \text{Re} \, a > 0 \tag{B.1}$$

and

$$\int d^2 z \, e^{-a|z|^2 + uz + v\bar{z}} = \frac{\pi}{a} e^{uv/a}, \quad \text{Re } a > 0; \ u, v \text{ complex}$$
(B.2)

which can be proven by elementary methods.

The gaussian integral

$$\int d^{2n} \mathbf{z} \, \exp\{-\langle \mathbf{z} | H | \mathbf{z} \rangle\} \tag{B.3}$$

is carried out by first noting that H, being Hermitian, is diagonalized by a unitary matrix U, namely $H = U^{\dagger}H_DU$, where $U^{\dagger} = U^{-1}$. We then perform the change of variables

$$|\mathbf{z}\rangle = U^{-1} H_D^{-1/2} |\mathbf{v}\rangle \tag{B.4}$$

so that $\langle \mathbf{z}|H|\mathbf{z}\rangle = \langle \mathbf{v}|\mathbf{v}\rangle = |\mathbf{v}|^2$, hence

$$\int d^{2n} \mathbf{z} \, \exp\{-\langle \mathbf{z} | H | \mathbf{z} \rangle\} = \frac{1}{\det H} \int d^{2n} \mathbf{v} \, e^{-|\mathbf{v}|^2} = \frac{\pi^n}{\det H}$$
(B.5)

The integral

$$L(\mathbf{k}, \mathbf{q}) \equiv \int d^{2n} \mathbf{z} \,\delta(\langle \mathbf{z} | H | \mathbf{z} \rangle - s) e^{-i(\langle \mathbf{k} | \mathbf{z} \rangle + \langle \mathbf{z} | \mathbf{q} \rangle)} \tag{B.6}$$

is carried out in similar fashion. Although in Sec. 7 we are only interested in real \mathbf{k} and \mathbf{q} , we can allow them to be, in general, complex vectors. The change of variables (B.4) yields

$$L(\mathbf{k}, \mathbf{q}) = \frac{1}{\det H} \int d^{2n} \mathbf{v} \,\delta(|\mathbf{v}|^2 - s) e^{-i(\langle \mathbf{k}' | \mathbf{v} \rangle + \langle \mathbf{v} | \mathbf{q}' \rangle)} \tag{B.7}$$

where $\langle \mathbf{k}' | = \langle \mathbf{k} | U^{-1} H_D^{-1/2}$ and $| \mathbf{q}' \rangle = H_D^{-1/2} U | \mathbf{q} \rangle$. Defining the 2*n*-dimensional real vector **V** and the 2*n*-dimensional complex vector **Q** as

$$\mathbf{V} = \begin{pmatrix} \mathbf{v}_x \\ \mathbf{v}_y \end{pmatrix}, \qquad \mathbf{Q} = \begin{pmatrix} \mathbf{k}'^* + \mathbf{q}' \\ i(\mathbf{k}'^* - \mathbf{q}') \end{pmatrix}$$
(B.8)

we obtain

$$L(\mathbf{k}, \mathbf{q}) = \frac{1}{\det H} \int d^{2n} \mathbf{V} \,\delta(V^2 - s) e^{-i\mathbf{V} \cdot \mathbf{Q}}$$
(B.9)

which is of the form (A.12), except that now \mathbf{Q} is complex. However, Eq. (A.13c) shows that $I(\mathbf{k})$ is an analytic function of q^2 , hence it is valid for complex q^2 as well. Therefore we obtain

$$L(\mathbf{k}, \mathbf{q}) = \frac{1}{\det H} \int d^{2n} \mathbf{V} \,\delta(V^2 - s) e^{-i\mathbf{V} \cdot \mathbf{Q}}$$
(B.10a)

$$= \frac{\pi^n}{\det H} \left(\frac{2\sqrt{s}}{Q}\right)^{n-1} J_{n-1}(Q\sqrt{s}) \tag{B.10b}$$

$$= \frac{\pi^n s^{n-1}}{\det H} \sum_{\ell=0}^{\infty} \frac{(-s\langle \mathbf{k} | H^{-1} | \mathbf{q} \rangle)^{\ell}}{\ell! \Gamma(n+\ell)}$$
(B.10c)

where Q is interpreted as $\sqrt{\mathbf{Q}^2}$, not as $|\mathbf{Q}|$. The final result, Eq. (B.10c) was obtained by substituting \mathbf{k}' and \mathbf{q}' in terms of \mathbf{k} and \mathbf{q} , *i.e.*,

$$Q^{2} = 4\mathbf{k}^{\prime *} \cdot \mathbf{q}^{\prime} = 4\langle \mathbf{k}^{\prime} | \mathbf{q}^{\prime} \rangle = 4\langle \mathbf{k} | H^{-1} | \mathbf{q} \rangle$$
(B.11)

No ambiguity should arise in the definition of Q in (B.10b) because it involves only even powers of Q.

$\begin{array}{cccccccccccccccccccccccccccccccccccc$	g(s)	g_ℓ	$A(n;\ell)$	$\mathcal{G}(\mathbf{k})$	comments
$\begin{split} s^{\alpha}e^{-s/2} & 2^{n/2+\alpha+\ell-1}\Gamma(n/2+\alpha+\ell) & \frac{(n/2+\alpha)\ell}{(n/2)\ell} & {}_{1}F_{1}(n/2+\alpha;n/2;-q^{2}/2) & \alpha > -\\ e^{-\sqrt{s}} & \Gamma(n+2\ell) & 2^{\ell}(\nu)\ell & (n/2+\alpha)/2 & \alpha > -\\ e^{-\sqrt{s}} & \Gamma(n+2\ell) & 2^{\ell}(\nu)\ell & 0 & (1+q^{2})^{-\nu} & \nu = (\\ \frac{1}{(s+1)p} & \frac{\Gamma(n/2+\ell)\Gamma(\mu-\ell)}{2\Gamma(n/2)} & \frac{1}{2^{\ell}(n/2)\ell} & \frac{1}{2^{\ell}(n/2)\ell} & 0 & (1+q^{2})^{-\nu} & \nu = \\ \delta(1-s) & \frac{1}{2\Gamma(n/2+\alpha+\ell)\Gamma(\beta+1)} & \frac{1}{2^{\ell}(n/2)\ell} & \frac{1}{2^{\ell}(n/2)\ell} & \Gamma(\nu+1)\left(\frac{2}{q}\right)^{\nu}J_{\nu}(q) & \nu = n \\ s^{\alpha}(1-s)^{\beta}\theta(1-s) & \frac{1}{2\Gamma(n/2+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2)\ell(n/2+\alpha+\beta+1)\ell} & 1F_{2}(n/2+\alpha;n/2,n/2+\alpha+\beta+1;-q^{2}/4) & \alpha > -\\ (1-s)^{\beta}\theta(1-s) & \frac{1}{2\Gamma(n/2+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2+\beta+1)\ell} & \frac{1}{2^{\ell}(n/2+\beta+1)\ell} & 0 & \nu = n \\ s^{\alpha}\theta(1-s) & \frac{1}{2\Gamma(n/2+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2+\beta+1)\ell} & \frac{1}{2^{\ell}(n/2+\beta+1)\ell} & 0 & \nu = n \\ s^{\alpha}\theta(1-s) & \frac{1}{2\Gamma(n/2+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2+\beta+1)\ell} & \frac{1}{2^{\ell}(n/2+\beta+1)\ell} & 0 & \nu = n \\ s^{\alpha}\theta(1-s) & \frac{1}{n+2\alpha} & \frac{1}{2^{\ell}(n/2+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2+\beta+1)\ell} & 0 & \nu = n \\ s^{\alpha}\theta(1-s) & \frac{1}{n+2\alpha} & \frac{1}{2^{\ell}(n/2+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2+\beta+1)\ell} & 0 & \nu = n \\ \frac{1}{2^{\ell}(n/2+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2+\beta+1)\ell} & \frac{1}{2^{\ell}(n/2+\beta+\ell+1)\ell} & 0 & \nu = n \\ \frac{1}{2^{\ell}(n/2+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2+\beta+\ell+1)\ell} & \frac{1}{2^{\ell}(n/2+\beta+\ell+1)\ell} & 0 & \nu = n \\ \frac{1}{2^{\ell}(n/2+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2+\beta+\ell+1)\ell} & \frac{1}{2^{\ell}(n/2+\beta+\ell+\ell+\ell+\ell+\ell+\ell+\ell+\ell+\ell+\ell+\ell+\ell+\ell+\ell+\ell+\ell+\ell+\ell+$	$e^{-s/2}$	$2^{n/2+\ell-1}\Gamma(n/2+\ell)$	1	$\exp(-q^2/2)$	(none)
$ \begin{split} e^{-\sqrt{s}} & \Gamma(n+2\ell) & 2^{\ell}(\nu)_{\ell} & (1+q^{2})^{-\nu} & \nu = (\\ \hline \frac{1}{(s+1)^{p}} & \frac{\Gamma(n/2+\ell)\Gamma(\mu-\ell)}{2\Gamma(p)} & \frac{\Gamma(\mu-\ell)}{2^{\ell}\Gamma(p)} & \frac{\Gamma(\mu-\ell)}{2^{\ell}\Gamma(\mu)} & (1+q^{2})^{\mu} K_{\mu}(q) & \ell < \mu \\ \delta(1-s) & \frac{1}{2} & \frac{1}{2} & \frac{1}{2^{\ell}(n/2)^{\ell}} & \Gamma(\nu+1) \left(\frac{2}{q}\right)^{\nu} J_{\nu}(q) & \nu = r \\ s^{\alpha}(1-s)^{\beta}\theta(1-s) & \frac{\Gamma(n/2+\alpha+\beta+\ell+1)}{2\Gamma(n/2+\alpha+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2+\alpha+\beta+1)\epsilon} & 1F_{2}(n/2+\alpha;n/2,n/2+\alpha+\beta+1;-q^{2}/4) & \alpha > r \\ \theta(1-s) & \frac{1}{n+2\ell} & \frac{1}{2^{\ell}(n/2+1)\epsilon} & \Gamma(\nu+1) \left(\frac{2}{q}\right)^{\nu} J_{\nu}(q) & \nu = r \\ (1-s)^{\beta}\theta(1-s) & \frac{1}{2\Gamma(n/2+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2+\beta+1)\epsilon} & \Gamma(\nu+1) \left(\frac{2}{q}\right)^{\nu} J_{\nu}(q) & \nu = r \\ s^{\alpha}\theta(1-s) & \frac{1}{2\Gamma(n/2+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2+\beta+1)\epsilon} & 1F_{2}(n/2+\alpha;n/2,n/2+\alpha+1;-q^{2}/4) & \alpha > r \\ s^{\alpha}\theta(1-s) & \frac{1}{n+2\alpha+2\ell} & \frac{1}{2^{\ell}(n/2+\beta+1)\epsilon} & \frac{1}{2^{\ell}(n/2+\beta+1)\epsilon} & 0 \\ \end{array}$	$s^{lpha}e^{-s/2}$	$2^{n/2+\alpha+\ell-1}\Gamma(n/2+\alpha+\ell)$	$\frac{(n/2+\alpha)_\ell}{(n/2)_\ell}$	$_{1}F_{1}(n/2+\alpha;n/2;-q^{2}/2)$	$\alpha > -n/2$
$ \begin{array}{cccc} \frac{1}{(s+1)p} & \frac{\Gamma(n/2+\ell)\Gamma(\mu-\ell)}{2\Gamma(p)} & \frac{\Gamma(\mu-\ell)}{2\Gamma(p)} & \frac{\Gamma(\mu-\ell)}{2^{\ell}\Gamma(\mu)} & \frac{\Gamma(\mu-\ell)}{2} & \frac{1}{\Gamma(\mu)} \left(\frac{q}{2}\right)^{\mu} K_{\mu}(q) & \ell < \mu \\ \delta(1-s) & \frac{1}{2} & \frac{1}{2\Gamma(n/2+\alpha+\beta)\Gamma(\beta+1)} & \frac{1}{2^{\ell}(n/2)\ell} & \Gamma(\nu+1) \left(\frac{q}{q}\right)^{\nu} J_{\nu}(q) & \nu = r \\ s^{\alpha}(1-s)^{\beta}\theta(1-s) & \frac{1}{2\Gamma(n/2+\alpha+\beta)\Gamma(\beta+1)} & \frac{1}{2^{\ell}(n/2)\ell(n/2+\alpha+\beta+1)\ell} & 1F_2(n/2+\alpha;n/2,n/2+\alpha+\beta+1;-q^2/4) & \alpha > r \\ \theta(1-s) & \frac{1}{n+2\ell} & \frac{1}{2^{\ell}(n/2+1)\ell} & \frac{1}{2^{\ell}(n/2+1)\ell} & \Gamma(\nu+1) \left(\frac{q}{q}\right)^{\nu} J_{\nu}(q) & \nu = r \\ (1-s)^{\beta}\theta(1-s) & \frac{1}{2\Gamma(n/2+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2+1)\ell} & \Gamma(\nu+1) \left(\frac{q}{q}\right)^{\nu} J_{\nu}(q) & \beta > r \\ s^{\alpha}\theta(1-s) & \frac{1}{n+2\alpha+2\ell} & \frac{1}{2^{\ell}(n/2+\beta+1)\ell} & \frac{1}{2^{\ell}(n/2+\beta+1)\ell} & 1F_2(n/2+\alpha;n/2,n/2+\alpha+1;-q^2/4) & \alpha > r \\ \end{array} $	$e^{-\sqrt{s}}$	$\Gamma(n+2\ell)$	$2^\ell(\nu)_\ell$	$(1+q^2)^{- u}$	$\nu = (n+1)/2$
$\begin{split} \delta(1-s) & \frac{1}{2} & \frac{1}{2} & \frac{1}{2^{\ell}(n/2)_{\ell}} & \Gamma(\nu+1) \left(\frac{2}{q}\right)^{\nu} J_{\nu}(q) & \nu = n \\ s^{\alpha}(1-s)^{\beta} \theta(1-s) & \frac{\Gamma(n/2+\alpha+\beta+\ell+1)}{2\Gamma(n/2+\alpha+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2)_{\ell}(n/2+\alpha+\beta+1)_{\ell}} & 1F_2(n/2+\alpha;n/2,n/2+\alpha+\beta+1;-q^2/4) & \alpha > - \\ \theta(1-s) & \frac{1}{n+2\ell} & \frac{1}{2^{\ell}(n/2+1)_{\ell}} & \Gamma(\nu+1) \left(\frac{2}{q}\right)^{\nu} J_{\nu}(q) & \nu = n \\ \theta(1-s) & \frac{1}{2\Gamma(n/2+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2+\beta+1)_{\ell}} & \Gamma(\nu+1) \left(\frac{2}{q}\right)^{\nu} J_{\nu}(q) & \beta > - \\ (1-s)^{\beta} \theta(1-s) & \frac{1}{2\Gamma(n/2+\beta+\ell+1)} & \frac{1}{2^{\ell}(n/2+\beta+1)_{\ell}} & \Gamma(\nu+1) \left(\frac{2}{q}\right)^{\nu} J_{\nu}(q) & \beta > - \\ s^{\alpha} \theta(1-s) & \frac{1}{n+2\alpha+2\ell} & \frac{n+2\alpha}{2^{\ell}(n/2+\beta+1)_{\ell}} & \frac{1}{2^{\ell}(n/2+\beta+1)_{\ell}} & 0 \\ \end{array}$	$\frac{1}{(s+1)^p}$	$\frac{\Gamma(n/2+\ell)\Gamma(\mu-\ell)}{2\Gamma(p)}$	$\frac{\Gamma(\mu-\ell)}{2^\ell \Gamma(\mu)}$	$rac{2}{\Gamma(\mu)}\left(rac{q}{2} ight)^{\mu}K_{\mu}(q)$	$\ell < \mu = p - n/2 > 0$
$s^{\alpha}(1-s)^{\beta}\theta(1-s) = \frac{\Gamma(n/2+\alpha+\beta)\Gamma(\beta+1)}{2\Gamma(n/2+\alpha+\beta+1)} = \frac{(n/2+\alpha)_{\ell}}{2^{\ell}(n/2)_{\ell}(n/2+\alpha+\beta+1)_{\ell}} = \frac{1}{1F_{2}(n/2+\alpha;n/2,n/2+\alpha+\beta+1;-q^{2}/4)} = \alpha > - \frac{1}{\alpha(1-s)} = \frac{1}{n+2\ell} = \frac{1}{2^{\ell}(n/2+1)_{\ell}} = \frac{1}{2^{\ell}(n/2+1)_{\ell}} = \frac{1}{2^{\ell}(n/2+1)_{\ell}} = \frac{1}{2^{\ell}(n/2+\beta+1)} = \frac{1}{2^{\ell}(n/2+\beta+1)_{\ell}} = \frac{1}{2^{\ell}(n/2+\beta+1)_{\ell}} = \frac{1}{2^{\ell}(n/2+\beta+1)_{\ell}} = \frac{1}{2^{\ell}(n/2+\beta+1)_{\ell}} = \frac{1}{2^{\ell}(n/2+\beta+1)} = \frac{1}{2^{\ell}(n/2+\beta+1)_{\ell}} = \frac{1}{2^{$	$\delta(1-s)$	2 <u>1</u> 1	$\frac{1}{2^{\ell}(n/2)_{\ell}}$	$\Gamma(u+1)\left(rac{2}{q} ight)^{ u}J_{ u}(q)$	$\nu=n/2-1$
$ \theta(1-s) \qquad \frac{1}{n+2\ell} \qquad \frac{1}{2^{\ell}(n/2+1)_{\ell}} \qquad \Gamma(\nu+1) \left(\frac{2}{q}\right)^{\nu} J_{\nu}(q) \qquad \nu = r $ $ (1-s)^{\beta} \theta(1-s) \qquad \frac{\Gamma(n/2+\ell)\Gamma(\beta+1)}{2\Gamma(n/2+\beta+\ell+1)} \qquad \frac{1}{2^{\ell}(n/2+\beta+1)_{\ell}} \qquad \Gamma(\nu+1) \left(\frac{2}{q}\right)^{\nu} J_{\nu}(q) \qquad \beta > r $ $ s^{\alpha} \theta(1-s) \qquad \frac{1}{n+2\alpha+2\ell} \qquad \frac{n+2\alpha}{2^{\ell}(n+2\alpha+2\ell)(n/2)_{\ell}} \qquad 1 P_2(n/2+\alpha;n/2,n/2+\alpha+1;-q^2/4) \qquad \alpha > r $	$s^lpha (1-s)^eta heta (1-s)$	$\frac{\Gamma(n/2+\alpha+\ell)\Gamma(\beta+1)}{2\Gamma(n/2+\alpha+\beta+\ell+1)}$	$\frac{(n/2+\alpha)_\ell}{2^\ell(n/2)_\ell(n/2+\alpha+\beta+1)_\ell}$	$_{1}F_{2}(n/2+lpha;n/2,n/2+lpha+eta+1;-q^{2}/4)$	$\alpha > -n/2, \beta > -1$
$(1-s)^{\beta}\theta(1-s) \qquad \frac{\Gamma(n/2+\ell)\Gamma(\beta+1)}{2\Gamma(n/2+\beta+\ell+1)} \qquad \frac{1}{2^{\ell}(n/2+\beta+1)_{\ell}} \qquad \Gamma(\nu+1)\left(\frac{2}{q}\right)^{\nu}J_{\nu}(q) \qquad \beta > - \frac{1}{s^{\alpha}\theta(1-s)} \qquad \frac{1}{\frac{1}{n+2\alpha+2\ell}} \qquad \Gamma(\nu+1)\left(\frac{2}{q}\right)^{\nu}J_{\nu}(q) \qquad \beta > - \frac{1}{s^{\alpha}\theta(1-s)} \qquad \frac{1}{s^{\alpha}\theta(1-s)} \qquad \frac{1}{s^{\alpha}\theta(1-s)} \qquad \Gamma(\nu+1)\left(\frac{2}{q}\right)^{\nu}J_{\nu}(q) \qquad \beta > - \frac{1}{s^{\alpha}\theta(1-s)} \qquad \frac{1}{s^{\alpha}\theta(1-s)} \qquad \frac{1}{s^{\alpha}\theta(1-s)} \qquad \Gamma(\nu+1)\left(\frac{2}{q}\right)^{\nu}J_{\nu}(q) \qquad \beta > - \frac{1}{s^{\alpha}\theta(1-s)} \qquad \frac{1}{s^{\alpha}$	heta(1-s)	$\frac{1}{n+2\ell}$	$\frac{1}{2^\ell (n/2+1)_\ell}$	$\Gamma(u+1)\left(rac{2}{q} ight)^{ u}J_{ u}(q)$	u=n/2
$s^{\alpha}\theta(1-s) = \frac{1}{\frac{n+2\alpha}{n+2\alpha+2\ell}} \frac{n+2\alpha}{\frac{2^{\ell}(n+2\alpha+2\ell)(n/2)_{s}}{2}} \frac{1}{1P_{2}(n/2+\alpha;n/2,n/2+\alpha+1;-q^{2}/4)} \alpha > -$	$(1-s)^{eta} heta(1-s)$	$\frac{\Gamma(n/2+\ell)\Gamma(\beta+1)}{2\Gamma(n/2+\beta+\ell+1)}$	$\frac{1}{2^\ell (n/2+\beta+1)_\ell}$	$\Gamma(u+1)\left(rac{2}{q} ight)^{ u}J_{ u}(q)$	$\beta>-1;\nu=n/2+\beta$
	$s^lpha heta(1-s)$	$\frac{1}{n+2\alpha+2\ell}$	$\frac{n+2\alpha}{2^\ell(n+2\alpha+2\ell)(n/2)_\ell}$	$_{1}F_{2}(n/2+\alpha;n/2,n/2+\alpha+1;-q^{2}/4)$	$\alpha > -n/2$

Table 1: Summary of examples.